

Lincoln University Digital Thesis

Copyright Statement

The digital copy of this thesis is protected by the Copyright Act 1994 (New Zealand).

This thesis may be consulted by you, provided you comply with the provisions of the Act and the following conditions of use:

- you will use the copy only for the purposes of research or private study
- you will recognise the author's right to be identified as the author of the thesis and due acknowledgement will be made to the author where appropriate
- you will obtain the author's permission before publishing any material from the thesis.

An Empirical Investigation of Credit Card Users in China: Evidence from Shijiazhuang Consumers

A thesis

submitted in partial fulfillment
of the requirements for the Degree of
Master of Commerce and Management

at

Lincoln University

by

Weikang Dong

Lincoln University, Canterbury, New Zealand 31_{th} March, 2014

Abstract

Abstract of a thesis submitted in partial fulfillment of the requirements for the Degree of M. C. M.

An Empirical Investigation of Credit Card Users in China: A Case Study of Shijiazhuang

By Weikang Dong

Since the first credit card issued by the Bank of China in 1985, the domestic banks has issued 140 million credit cards in 2008, and by the end of 2009, the number of credit cards issued by the domestic banks reached 190 million, an increase of 30.37% compared to 2008. The domestic credit cards transaction totaled 2.1 trillion RMB in 2008, and this number reached 3.5 trillion RMB in 2009, an increase of 69.9% from 2008 (CIW Team Staff, 2012). At the end of 2011, domestic banks in China have issued 285 million credit cards (Peng, 2012). Today 30 percent of Chinese urban households own at least one credit card and the growth rate of credit card adoption has been an average of 40 percent per year between 2004 and 2009 (Hurst, 2012).

This study used binary logit and ordered logit models to identify and examine the factors influencing consumers' use of credit card and the choice of different level of credit limit in China. More importantly this study identifies why consumers use and do not use credit card in China. The data is collect in Shijiazhuang city, the capital city of Hebei Provence.

Further, the finding shows consumers who are married and in the age group of 18 to

35 years old are the major credit card users.

Key words: credit card, China, electronic transaction, payment

iii

Acknowledgement

I would like to express my most heartfelt thanks and appreciation to my main supervisor Dr Christopher Gan a respectable, responsible and resourceful scholar, who consistently assisted my thesis work, especially in data collection and analysis of the research findings. Thank you also to my associate supervisor, Dr Baiding Hu, who assisted me in my statistical analysis. I would like to thank them for their patience, valuable guidance, and knowledge throughout the research. Without their exceptional support and encouragement, I could not have completed my thesis.

I also would like to thank all my friends who have assisted and supported me while writing my thesis. I would like to thank Nhung who assisted me on the application of SPSS and data collection process. My appreciation also goes to my postgraduate fellows and staff in the Faculty of Commerce for their support and assistance.

Finally, thanks and love in abundance to my parents, who provided much support, both financially and emotionally. Thanks for their endless love and support that guided me through the hard times when I was studying away from home.

Contents

Abstract	ii
Acknowledgement	iv
CHAPTER ONE	1
1. Background of Chinese Credit Card	1
1.2 Problem Statement	6
1.3 Research Objectives	7
1.4 Structure of the Thesis	7
CHAPTER TWO	9
2. Introduction	9
2.1 General Credit Card Market	9
2.2 China's Economy Development	14
2.3 China's Credit Card Market	16
2.4 Factors Impacting Credit Card Use	21
2.5 Features of Credit Card	29
2.6 Demographic Factors	32
2.7 Summary	34
CHAPTER THREE	35
3. Introduction	35
3.1 Empirical Framework	35
3.2 Research Model	37
3.3 Questionnaire Development and Data	47
3.4 Data	49
3.5 Sampling Method	50
3.6 Data Collection Method	50
3.7 Sample Size	51
3.8 Summary	52
CHAPTER FOUR	53
4. Introduction	53

4.1 Descriptive Frequencies of the Respondents	53
4.2 Assessment of the Data	55
4.3 Correlation Coefficients of Explanatory Variables in Each Model	55
4.4 Data Level5	56
4.5 Empirical Results and Findings of the Research	56
4.5.1 Results Pertaining to Research Objective One	57
4.5.2 Results Pertaining to Research Objective Two	67
4.5.3 Results Pertaining to Research Objective Three	75
4.6 Marginal Effect Analysis	76
4.7 Summary 8	80
CHAPTER FIVE	81
5. Introduction	81
5.1 Overview and Summary	81
5.2 Conclusions Pertaining to Research Objective One	82
5.3 Conclusions Pertaining to Research Objective Two	84
5.4 Conclusions Pertaining to Research Objective Three	85
5.5 Implications	86
5.6 Limitation and Recommendation for Future Research	88
5.7 Conclusions	89
References	90
Appendix One: Tables	97
Appendix Two: SURVEY QUESTIONNAIRE COVER LETTER 11	15

List of Table

Table 1.1: Comparison of Five Years Real GDP Growth of China, US and UK	1
Table 1.2: Credit Card Growth in China from 2008 to 2011	2
Table 2.1: Card Brands and Transactions Values in Asia Pacific 2007	12
Table 2.2: Total Value of Transaction (General Purchase) by Country	12
Table 3.1: Variables Definitions used in Equation (3.5)	40
Table 3.2 Variables Definitions in used in Equation (3.8)	44
Table 3.3: Variables Definitions used in Equation (3.9)	46
Table 4.1: Empirical Result (Logit Model of Credit Card Users versus Non-Credit Card	
Users)	59
Table 4.2: Result for Research Objective One	64
Table 4.3: Mean Values of Credit Card Attributes Measurement (All Respondent)	65
Table 4.4: Characteristics for Not Using Credit Card	66
Table 4.5: Empirical Results (Ordered Model of Credit Card Users)	69
Table 4.6: Marginal Effect of Each Outcome of Ordered Logit.	70
Table 4.7: Result of Research Objective Two	74
Table 4.8: Result for Research Objective Three	76
Table 4.9: Marginal Effects of Credit Card Users and Non-Credit Card Users	77
Table 4.10: Marginal Effect of Factors Impact Consumers Decision to Use Credit Card	78
Table 4.11: Marginal Effect of Demographic Factors of Credit Card Users and Non-Credit C	Card
Users	79
Table F.1: Descriptive Statistics of Socio-Economic Characteristics of All Respondents	97
Table F.1.1: Profile of Surveyed Respondents	99
Table F.2: Descriptive Statistics of the Respondents (Credit card user versus Non-credit	card
user)	102
Table F.3: Descriptive Statistic of the Respondents' Socio-Economic Factors.	104
Table F.4: Statements Pertaining to the Respondents' Attitude toward Credit Cards	104
Table F.5: Important Factor for Respondent to own a Credit Card with 1 (least important) to	8
(most important)	108
Table F.6: Importance of the Loss of Financial Control When Choosing Credit Card	110
Table F.7: Correlation Coefficient of Objective One and Two	111
Table F.8: Correlation Coefficient of Objective Two	113

CHAPTER ONE INTRODUCTION

1. Background of Chinese Credit Card

Economic growth in China has been impressive and if the trends stay as they are now it will surpass the US as the biggest GDP country in the medium term (Vilaclara, 2009). Table 1 shows the real GDP growth of China compared with US and UK from 2007 to 2011.

Table 1.1: Comparison of Five Years Real GDP Growth of China, US and UK

Years Countries	2007	2008	2009	2010	2011
China	14.2%	9.6%	9.2%	10.4%	9.3%
US	1.9%	-0.4%	-3.5%	3.0%	1.7%
UK	3.6%	-1.0%	-4.0%	1.8%	0.8%

Source: The World Bank (2013)

China exhibited a strong growth in the real GDP in 2008 and 2009 while U.S and U.K exhibited negative GDP growth. This reflects China's fast and strong growing economy which improves people's living standard, and consumption level. This gives rise to the emergence use of credit card, which brings great convenience to people having to carry large amounts of cash. Thus, people have started to use credit card more frequently and there has been an increased in the number of credit cards issued by commercial banks. For example, Table 2 shows the growth of total credit card payment, the total number of credit card transaction and credit card issued in

China from 2008 to 2011, which reflected a strong and continuous growth in the use of credit card in China.

Table 1.2: Credit Card Growth in China from 2008 to 2011

Years	2008	2009	2010	2011
Credit card total payment (Trillion Yuan)	2.1	3.5	5.1	7.6
Total No. of credit card transactions (Billion)	1.49	1.97	2.4	2.85
Total credit card issued (Million)	140	190	230	290

Source: CIW Team Staff (2012)

In 1985, Bank of China launched the first credit card in China. From then, China start to develop the credit card market, but compare to the western country, the development of credit card in China is just at the developing stage. China became a member of World Trade Organization (WTO) in 2001 and started to liberalize the financial industry in the beginning of 2006, including the credit card industry. The opening up of the financial industry allows foreign banks to enter into the credit card industry in China, which serve not only as a vehicle for payment transactions, but also as a means of borrowing. Chinese domestic commercial banks felt the pressure and began to pay more attention to the credit card industry. Chinese domestic commercial banks have been using China UnionPay to aggressively promote their credit cards and amassed a large number of loyal customers (Gao, 2011).

China UnionPay was launched in Shanghai by People's Bank of China in 2002.

China UnionPay is the only domestic bank card organization in China, and is the

only inter-bank network in mainland China that links the ATMs of both major and small banks in China. China UnionPay allows the credit cards issued by domestic banks to be used in different regions and different countries. In New Zealand, China UnionPay has been accepted by the Bank of New Zealand (BNZ), where UnionPay credit card holders can use their card directly at any BNZ ATM. Further, over 7700 retailers in New Zealand will accept UnionPay credit card payment (BNZ, n.d.). Today China UnionPay credit card is also affiliated with American Express, MasterCard or Visa, and they can be used as an American Express, MasterCard or Visa (Moses, 2010). One of the goals of China UnionPay is to create an independent bankcard brand of China by cooperating with China domestic commercial banks to push forward the sustainable and healthy development of China's bankcard industry (Vilaclara, 2009).

Credit card as a bank credit instrument acts as a tool of payment which improves the efficiency of transaction payment, reduce transaction costs and improve the security of transactions (Wenku, 2013). Today, credit card plays an important role in people's daily lives, it has made people's consumption of goods and services more easily and the proportion of people using credit card to purchase goods and services increases every year. According to the data from CIW Team Staff (2012), there were about 290 million credit cards issued in 2011 in China, an increase of 24.3% compare to 2010; the total credit card transactions in 2011 totaled 7.56 trillion RMB, an increase

The credit card holders in China have increased rapidly. The growth of credit card usage can be expected to come from the increasing number of young affluent Chinese. Young people are innovative, creative, willing to take risk and thus are expected to try new things. They are influenced less by the traditional Chinese culture such as saves first spend later. Young affluent Chinese have a positive attitude toward having a credit card and see the benefits of the credit card when travelling overseas and easier to pay by credit (Thompson, Worthington & Stewart, 2009). Thompson, et al.'s study reveals that for the young affluent Chinese, the holding of credit cards is widespread, with some students having more than one credit card. Their finding demonstrated that credit card usage levels are influenced by the ease of use and the credit card acceptance service provided, and the young affluent are more likely to avoid carrying cash.

Credit card gives people lots of convenience on purchasing goods and services, since they do not need to carry cash, but the Chinese consumers are still influenced by the traditional concept of spending, especially among the older Chinese people, which is saved first and then spend. China's households are very savings centric; the average Chinese family saves 25 per cent of their discretionary income, about six times the saving rate for US households (Thompsonand Worthington, n.d.). Another

factor that will influence the Chinese consumer is the income level. Since consumer needs to repay their debts, the income level will dictate whether they can afford to have a credit card or not.

Another reason why Chinese people do not use credit card extensively is the safety reason. This is especially true for online payment, since they do not need to enter a password when using the credit card to make a payment online. Further, in some transactions, the consumers just need to show their signatures to the cashier rather than entering password when using a credit card. This makes it is very easy for other people to use a lost or stolen credit card. Feldman's study (2007) argues that consumers will face the perceived risks and dangers as they release their credit card information into cyberspace. When given the choice, many customers would rather visit a shop or order something over the phone because hackers have and probably will find ways to steal credit card numbers from websites. Thus, many people prefer not to use credit card especially in China.

Consumers in China choose not to use credit cards because there is less acceptance point of the credit card in many retail stores. There are about 14 million merchants in China and only 300,000 accept credit card, which is only 3% of the merchants who accept credit card transactions. The difficulty of applying a credit card can be another reason why many Chinese people do not use credit cards. A report by

MasterCard and China Women's News shows 60% of the respondents said it is very difficult to apply for a credit card, such as providing many supporting documents and filling up more forms, and also the need to wait for an extended time to get the credit card from the bank (Sina, 2001).

This study identifies and examines the factor that influence consumers' use of credit card in China, and more importantly, this study identifies why Chinese consumers use or do not use credit card.

1.2 Problem Statement

For retail transactions, consumers have several choices of payment instruments, including cash, check, credit cards, and debit cards. Each choice provides a host of desired properties that differentiate one instrument from the other. The overall aim of this study is to investigate which group of people is the major credit card user in China. There are limited empirical research studies on the user behavior of the credit card in China. With a large population and the opening up of the financial industry, China has a huge potential market for credit cards. Further, the living standard in China has improved since China opened to the world, and Chinese people are influenced by some Western cultures. The new generations of Chinese youths see traditional Chinese culture as being old fashioned or obsolete while Western culture is accepted as the current trend. The American modes of doing things becomes a

model to the young generation of the Chinese society, and this bring a lot of transformation into the Chinese economy and life style because they adopt the Western values with Chinese modification (Odinye & Odinye, 2012). Today, Chinese people are more open minded and practice less traditional consumption concept. Thus, it is important to understand what factors influence Chinese consumers to use credit card and why some do not use credit cards. The findings of this study can help the banks to have a clear direction of which group of people they should target to promote their credit cards and what type of factors impact (or influence) people use of credit cards. This enables the banks to come up with more effective promotional campaigns on credit cards.

1.3 Research Objectives

The research objectives include:

- 1) To identify the factors that influence consumers' decision to use credit cards.
- 2) To determine the factors that impact credit card users' choice of different level of credit card limit.
- 3) To examine whether different demographic characteristics impact credit card users versus non-credit card users.

1.4 Structure of the Thesis

Chapter One provides an overview of the research background, research problem statement and objectives. Chapter Two reviews the literatures on the development of

credit card and the current credit card market in China. Further, it reviews different factor that impact credit card use, followed by the features of credit card and the credit card usage among consumers in different demographics groups. Chapter Three explains the data collection, variables selection, and methodology used in the study. Following this, Chapter Four presents the discussion of the empirical results and findings, and Chapter Five provides the conclusions of research findings, policy implications, limitations and recommendations for future research.

CHAPTER TWO LITERATURE REVIEW

2. Introduction

This chapter reviews the literature on credit cards. This chapter is organized as follows. Section 1 provides a brief overview of the development of credit card. Section 2 discusses the economic development in China and how it is related to the credit card use. Section 3 discusses the Chinese credit card market and the reason why China will become the biggest credit card market in the world. Section 4 discusses the factors that might influence the consumers' decision whether or not to use credit card. Section 5 discusses the advantages of credit card. Section 6 discusses the demographic of the credit card users in China.

2.1 General Credit Card Market

Today, credit card has become an extremely popular payment instrument. The popularity of credit cards continues to grow as evidenced by a greater proportion of merchants that accept them and consumers that carry them (Chakravorti, 2006). The primary use of credit card today is a transactional medium, not as a source of credit. Garcia (1980) also states that when using credit card to purchase, consumers will be allowed to settle their credit card payment bill after a period of time. In the case of store cards, the retailer provides the credit, in the case of bank, gas and travel and entertainment cards, the retailer received payment from the card-issuer, who bills the consumers at the end of the month. Holders of gas, retail, and bank credit cards

usually have the option to revolve credit and pay interest, typically at the rate of 18 present per annum, or to repay before the next billing date and avoid interest charges (Garcia, 1980). Over half and probably as much as 68% of credit card users should be considered "convenience users", who use credit cards primarily as a transactional medium and who pay off their balance in full each month (Zywichi, n.d.).

Credit cards were first issued in the United States by hotels. During the 1960's, the BankAmericard, separate into Visa and the Master Charge systems, eliminated competition and established themselves, effectively, as the bank credit card industry (Garcia, 1980).

From 1967 to 1977 in United States, the credit card use in general has grown at an average annual rate of 12.2 persent. This growth has been faster than growth of the consumer expenditures (9.0%), and total instalment credit (10.0%). Since data first became available in 1967, the highest growth rates of the credit card usage have occurred during the expansionary phases of the business cycle in 1973 and 1977 (Garcia, 1978). Further, when credit is tight elsewhere, consumers may substitute their credit card line of credit for more traditional instalment loans (Peterson 1976).

Today credit card serves as an indispensable credit and payment instrument in the US (Chakravorti, 2006) and becomes the second most popular non-cash instrument

and growing in popularity around the world (Chakravorit, 2003). For example, in 2012, the estimated number of US credit card holders is 181 million and the estimated number of credit cards is 609.8 million. This means there are many people in US hold more than one credit card, and the annual purchase volume of credit cards in 2011 was \$2.1 trillion. Further in 2012, two-third of Americans owned at least one credit card (Infographixdirectory, 2012).

Asia Pacific has not always been at the forefront of innovations in card payment transactions, but changes have occurred across the regions significantly. Economies of scale have increased as a result of growth in credit card usage and now are dominated by debit cards and cash (KPMG International, 2009).

At the end of 2007, credit card payment in the Asia Pacific region was close to USD 1.3 trillion, around 30 percent of the global figure. The region has about 50% of the world population, this means card usage in the Asia Pacific region has a large growth potential. There are nearly 70% of the card usages for the purchase of goods and services, and around 30% for forwarding or the withdrawal of cash. Visa and MasterCard accounted for 90 percent of the card usage by volume and by values (see table 2.1) (KPMG International, 2009).

Table 2.1: Card Brands and Transactions Values in Asia Pacific 2007

Card brand	Cards (million)	Transactions Value (\$ billions)		
		Total	Purchases	Cash
Visa	446	845	556	289
MasterCard	189	311	215	96
JCB	59	67	62	5
American Express	10	55	54	1
Diners Club	3	11	10	1
Total	707	1,289	897	392

Source: KPMG International (2009)

Table 2.2 shows Japan leads in the ranking of total transactions, with USD 209 billion of purchases, where over 99% of transaction in Japan involves credit cards, followed by South Korea. China and India exhibited large growth potential with USD 24 billion and USD 2 billion of card payment, respectively when considered the large population and economic growth of both countries.

Table 2.2: Total Value of Transaction (General Purchase) by Country

Country	Total Value of Purchases by Card
Japan	USA 209 billion
South Korea	USD 203 billion
Australia	USD 140 billion
Taiwan	USD 24 billion
China	USD 24 billion
Hong Kong	USD 20 billion
New Zealand	USD 11 billion
Thailand	USD 6 billion
India	USD 2 billion

Source: KPMG International (2009)

In Australia, card payments market has been growing steadily. By the end of 2008,

with a population of 21 million, there were over 14 million credit and charge card accounts in Australia, and this number has rose to 16 million in circulation in 2010. Further, in October of 2009, Australians spent \$19.89 billion on credit cards. In the UK, a country of about 60 million people has about 50 million credit cards in circulation in 2009. Russia is another country that has rapid rise in the number of credit cards in the past few years. In 2007, there were just 12 million credit cards in Russia but by 2009 that figure had risen to 24 million and analysts predict there will be more than 35 million credit cards in use by the end of 2011 (Credit Card Bible, 2011).

Credit card usage in China has also increased dramatically over the last few years. In 2003, there were only 3 million credit cards in circulation. In November 2008, the number of credit cards held by Chinese citizens had risen to 160 million. Many people admit to holding multiple credit cards and regularly use them to pay bills such as rent and mortgage debts as well as for retail spending (Credit Card Bible, 2011). All RMB-denominated cards issued in China run through the China UnionPay (CUP) scheme, regardless whether it is credit card or debt card. Since its inception in 2002, CUP has also operated the key payments network in China, providing an inter-bank and intra-country card network that enables interoperability between different banks and cards. In 2004, China UnionPay credit card can be used in Hong Kong and Macau. In 2005, Thailand, South Korea and Singapore retailers

started to accept China UnionPay credit card (Wenku, 2007). The widely acceptance of China UnionPay in the foreign countries bring convenience to the UnionPay card holders when traveling overseas, and avoid the inconvenience of carrying cash and minimize the exchange of rate volatility. Based on the overview of development of the credit card in the past, credit card has become an important part of people's daily life.

2.2 China's Economy Development

China's economy has developed rapidly in the past few years. Most of this growth has come from higher labor productivity. From 1978 to 2011, the real GDP (grow domestic product) growth averaged about 10% per year (Haltmaier, 2013). According to the data from People's Daily Paper (2013), China's GDP in 2012 is 103 times more than the GDP in 1979, while its total economy in the world share has grown from 1.8% in 1979 to 11.5% in 2012. The fast growth of China's economy in the past 35 years is evidenced by the following factors: (1) The GDP has grown from 364.5 billion RMB in 1978 to 51.9 trillion RMB in 2012; (2) The total economy ranking in the world has risen from tenth in 1978 to second in 2010; (3) Per capita GDP has risen from 381 RMB in 1978 to 38,420 RMB in 2012 (People's Daily Paper, 2013).

The living standards and the income of Chinese people have also improved.

According to an analysis of official government statistics by China Market Research

Group, the average disposable income of urban Chinese households rose to around \$3,000 per capita in 2010, but in 2000, the income was just \$760 per capita (Censky, 2012). In 2012, the average urban Chinese households' income was 26,959 RMB, the average disposable income of urban Chinese households income was 24,565 RMB, and the real growth rate was 9.6% compare to 2011 (McKinsey&Company, 2012). Further, according to the World Bank (2012), the Gross National Income per capita has grown from \$190 in 1978 to \$5,680 in 2012. Thus, China has jumped from the lower income country to the upper-middle income country. Higher income means people have more money to purchase goods and services and the ability to apply for credit cards.

As people's living standards improve, the consumer market also change over time, the most significant change is the Chinese consumers' behaviors becoming more similar to the consumers in the developed countries (McKinsey&Company, 2012). For example, instead of saving first then spending, they are more likely to use credit card to incur overdraft consumptions (Wang, 2013). This is especially so for the young people where they are more likely to use credit card to overdraft when they do not have enough money in their hands (Ma, 2013).

Household consumption has grown rapidly in China over the past two decades, averaging about 8% a year and rising to 10% in the past few years (Baker &

Orsmond, 2010). But the Chinese consumption is only one-third of the GDP, while in U.S., the consumption to GDP ratio is above two-third.

In the recent years, the growth of China economy has started to slow down, averaging around 8% growth each year, but, the consumers' consumption power is still growing rapidly. This can be seen from the higher disposable income in most urban household in China. It is forecasted that in 2020, the household average annual disposable income will account for 57% of all urban household (McKinsey&Company, 2012).

Since the growth of the country's per capita GDP and the living standard has been improved, people earn more money than before, and more willing to spend on luxury goods and/or leisure, such as new clothes, dining out and better electric devises. These encourage the consumers to use credit cards because of convenience and they do not have to carrying lots of cash.

2.3 China's Credit Card Market

At the beginning of 2006, credit card users in China soared to 12 million from just slightly 3 million at the beginning of 2003 (Tan, 2006) "When it comes to credit cards, China is a virgin market with vast opportunity," says Michael Lafferty, chairman of U.K. based Lafferty Group, who predicts that China will become the

world's largest credit card market by 2015 (Chu, 2012). Chinese credit card market is largely directed by political calculation, policy mission and administrative intervention. Even though economic and social factors play a vital role, government regulations and policies control or strongly influence the competitive landscape, the credit card ecosystem and the industry income structure. Even so, the Chinese credit card market will continue to be the most promising market worldwide due to the size of the national economy and population.

The Bank of China issued the first credit card in 1985 and the development of China's credit card can be divided into four stages (Zhou, 2010).

i. First stage (1985 – 1994):

The credit card in China was more likely a Quasi-credit card, and the application process was very difficult, the highest credit limit at that time was 5,000 RMB. Also, the average income of the Chinese people and consumption power were very low, where the banks issued only 8 million credit cards.

ii. Second stage (1995 – 1999)

The credit card industry came to a standstill from 1995 to 1999. During this period, many banks started to issue debit cards, which hampered the issuance of quasi-credit cards. This action almost brought the development of the credit card at a standstill. At the end of 1999, there were only 18.5 million credit cards issued in China, but there were more than 100 million debit cards issued by the banks.

iii. Third stage (2000 – 2002)

China's credit card industry began to develop comprehensively during this period, where the People's Bank of China and relevant departments started to pay more attention to the credit card industry. In March 2002, China UnionPay was established, where it is the only inter-bank network in mainland China that links the ATMs of both major and small banks in China. It allows the credit cards issued by domestic banks to be used in different regions and different countries. This is a millstone in the development of China's credit card industry. The international credit card organizations also have played a very important role on the development of China's bankcard industry. MasterCard and Visa Card are first to support the international standard dual-currency and single-currency credit card issued by the banks in China. The dual-currency credit card include both the RMB-denominated account in conjunction with China UnionPay, and a foreign-currency-denominated account, in conjunction with the foreign scheme, such as Visa. This means if the card is used in China, it is treated as a RMB transaction, processed through the China UnionPay network. If the card is used abroad, it is settled in the currency of the card, and processed through the foreign scheme's international network. China UnionPay is working to expand acceptance of its cards in the international market. For example, in 2013, there are 142 counties and regions accepting UnionPay cards (UnionPay, 2013). In New Zealand, UnionPay cardholders can use their card with Bank of New Zealand (BNZ) ATM and there are over seven thousand retailers

accepting UnionPay card payment (BNZ, n.d.).

iv. Fourth stage (2003 to present)

The credit card industry entered a competitive development stage, where each bank in China has set up its own credit card center and started to launch the China UnionPay standard credit card.

The credit card industry is the main part of the national economy, where it plays an important role in the growth of China economy and the growth in GDP. At the end of 2009, domestic credit card transaction proportion of total retail sales of social consumer goods has risen from 11.2% in 2007 to 27.9% in 2009 and this ratio has risen to 48.26% in 2012, which means nearly half of the transactions are paid by credit cards. The credit card transaction to GDP ratio has risen from 4% in 2007 to 10.4% in 2009 (Zhou, 2010).

In the term of reducing social transaction cost, credit card shows a significant advantage. In 2007, the social transaction cost of credit card transaction volume ratio was 0.41%, while the cost of cash transaction ratio was 1.76% (Zhou, 2010). In 2011, the credit card has saved up to 102.1 billion RMB in the transaction cost (Peng, 2012).

At the end of 2011, the total number of credit card issued reached 285 million, it

increased 24.3% from 2010, and 79.41% of the consumers are have no more than 3 credit cards, 35.12% have only one credit card while 1.35% have more than 10 credit cards. The total transaction reached 756 million RMB, which was 47.95% higher than the transaction volume in 2010. The number of the domestic credit card merchants has increased at the end of 2011 and the number of domestic acceptant merchants reached 3.18 million, a 45.68% increase compared to 2010 (Peng, 2012).

Vilaclara (2009) identified four reasons why China is a big potential market for the credit card business. First, there is not many people using credit card in China at present, the number of cards hold per person is 0.18 in China. Compared to U.S., there are more than 8 credit cards per person on average. Second, China is a market where credit is not used as often as in U.S. and the Chinese consumers are more savings oriented than U.S.. The average unpaid balance, credit left revolving in the cards instead of being used is only 2.5 billion RMB. If this trend is similar to the more developed economies there will be a very big growth on this numbers as consumers start to use the credit cards to carry balances. Third, as of 2007, the Chinese domestic banks have not made profit in the credit card business. However, the banks have been aggressive in the credit card business recently because they see potential profit making in the futures. For the multinational banks operating in China, the outlook is a bit different as they are already making profits on their credit card business. This is probably because the customers on average are more sophisticated.

Fourth, if we use Hong Kong' credit card market as a reference point where mainland China wants to follow, we can see that 81% of the household in Hong Kong own a credit card at the end of 2007, but in mainland China, there are only 14% of household owning a credit card. In addition, the big promotion of credit cards by Chinese banks has turned the credit card market into a big and profitable business.

2.4 Factors Impacting Credit Card Use

Ando & Modigliani's (1963) life cycle model assumed that consumer will try to maximize their utility from their lifetime consumption. This model implies that the consumers will spend more than their household income in their lifetime through borrowing.

Bryant (1990) explained that borrowing is the transfer of future resources to the present to increase current consumption. Under the budget constraint and the consumption needs, consumers will choose to borrow to maximize utility, and the credit card can be an easy way to borrow money to make consumptions. Differences in consumers' income, age and household size, marital status, education and interest rate may influence a consumer willingness to own and use a credit card.

Age: Previous studies such as Bei (1993), Canner & Cynak (1985), Choi &

DeVaney (1995), Steidle (1994) and Wasberg, Hira & Fanslow (1992) find there is a negative relationship between age and being a credit card revolver. This finding implies that younger households are more likely to use credit card as a borrowing instruments than older households (Kim & DeVaney, 2001). Awh & Waters (1974) also find that older age reduces the likelihood of an individual being an active credit cardholder. This is because older people's spending patterns probably do not need a significant amount of credit thus reduces the use of credit cards. Further the authors argued that older people tend to resist innovations and those who have grown up in a computerized society may be less resistant to such banking innovation as credit card (White, 1975). White's study showed that older people tend to use credit card less as they have less opportunity to use the credit card then younger people.

Gender: Armstrong & Craven (1993) find that gender was a predictor of how many credit cards consumer hold, and they find that females tend to have a higher average number of credit cards than males. However, Khare et al. (2011) find that males are more likely to own more credit cards and use them as a form of revolving credit than female. Similarly, Hirschman & Goldstucker (1977) find that bank credit card users were more likely to be males. The different result in credit card use between females and males can be explained that the female and male have different shopping behaviour and they may use credit card in different ways. For example, Kaynak & Harcar (2001) find women tend to use credit cards to purchase household goods,

clothing and personal belongings while men use their credit card to pay for electronics, entertainment, travel and food. Markovich & DeVaney (1997) investigate the credit card use in the college students, they found that seniors with four or more credit cards had more debt than the seniors with one to three credit cards, but there was no mention about the gender differences. White (1975) find that male probably find credit cards relatively more convenient to use when making purchase than female when they make their purchases.

Marital Status: Marital status also impact credit card use. Canner & Cyrnak (1985) show that marital status is important in explaining who is a revolving credit card user. Kinsey (1981) and Steifle (1994) find that consumers who are married have higher expenditures than the consumers who are not married. Higher expenditures means the consumers need to spend more money or borrow, and the credit card can be the easiest way for consumers to borrow to purchase goods.

Yieh (1996) find that households headed by individuals who were female, married and unemployed were more likely to have a negative attitude toward installment borrowing. This implies such women will prefer not to use the credit card because they may have lower income but will have higher expenditures. They might not have the ability to pay off the credit card bill each month and choose not to get into debts.

Household Size: Household size is another factor that influences the consumers to be a credit card user. Godwin (1998) showed there is a positive relationship between the household size and the increase in the household debt. Similarly, Kinsey (1981) find that two-people households have more credit cards than single-person household as two-people households tend to have higher expenditures in their daily life. This finding supports the demand for the present consumption is positively related with the household size. Thus, the needs of a large amount of living expenses can be reflected in borrowing money from credit card. This is consistent with Chien & Devaney's (2001) study where household with a large household size are more likely to have favourable specific attitudes toward using credit card and higher outstanding credit card balances.

Among the households that have a credit card, Bertaout & Haliassos (2001) found that households with more children are more likely to revolve credit card debts because more children reflect the households' higher expenditure in their daily life.

Education: Education is considered as one of the human resources and can increase the demand of current consumption (Kim & DeVaney, 2001). According to Becker (1975), education can be a future resource, as well as a current human resource. Higher resources mean higher income in the future, and the higher income is likely to increase the demand for consumption and borrowing in the present (Kim &

DeVaney, 2001). This will make people more likely to have a credit card since it is an easy form of borrowing. This is consistent with Barker & sckerkaya (1992), Chien & Devaney (2001) and Choi & DeVaney (1995) studies where they found the education level was positively related to credit card use and the people with more education were more likely to have favourable specific attitudes toward credit card use and tend to use credit card more often.

Income: The effect of net worth is similar to the effect of current income, which can determine a household's level of consumption (Bryant, 1990). If net worth is constrained, consumers can borrow money from credit cards to meet their needs (Kim & DeVaney, 2001). Upper-income consumers exhibit more favorable attitudes toward credit cards than lower-income consumers (Mathews & Slocum 1972, Slocum & Mathews 1970). This is because the lower income consumers may not have the ability or money to pay off the bill on credit card when due.

Mathews & Slocum (1969) and Slocum & Mathews (1970) find that cardholders with low income and low socioeconomic status are more likely to use one credit card to repay the other credit card bill. This is because they cannot repay their credit card bill using their income only.

Also for the credit card user, Steidle (1994) find that income was negatively related to

credit card debt. This means higher income people have lower credit card debt, since higher income people can pay off their credit card debt on time, or they may not use the credit card frequently since they have enough money to purchase the goods instead of using the credit card.

Mandell (1972) find that in comparison with higher income families, families earning less than \$10,000 per year were much more likely to maintain high outstanding credit card balances and to treat those balances as a type of instalment debt.

In Monger's (1992) study, it reported that people who were married, with higher socio-economic status and higher level of educations have multiple card accounts. This implies that the people who are wealthy are using their credit cards for convenience instead of revolving the debt each month as they have the ability to pay off the credit card bill each month.

Interest rate: The interest rate and the payment pattern ultimately determine the credit card purchases and balances (Duca & Whitesell, 1995). Canner & Luckett (1992) found that credit card revolvers were more likely to be sensitive to the level of interest rate. This is because when credit card users cannot pay off their credit card bill before due day, they will end up paying higher interest rate to the bank. Higher interest rate means the users will pay more than they actually spend on their

credit card when they cannot pay off their credit card bill on time. Higher interest rate is also important for the low income consumers to consider whether to use the credit card or not, as low income means the probability of not paying off the credit card bill on time is greater.

The interest rate will activate after the credit card user cannot pay off the credit card bill on due day. For some people who have higher income interest rate on credit card is not a big issue since they can pay off the credit card bill on time and have less credit card debt (Steidle, 1994).

In China, the credit card usually has a 25 days grace period (Hexun, 2009). This means even if the consumers pay off their credit card bill after the due date but within 25 days, they still do not need to pay the credit card interest. This gives the consumers more time to prepare and pay off their credit card bills and reduces the need to pay the credit card interest.

Occupation: Based on the studies of Hong & Beak (2004) and Kinsey (1981) people who are employed in unskilled jobs and those who are generally unemployed are significantly less likely to have credit card than those employed people. And people who are in professional occupations are no more likely than those in blue-collar jobs to have credit card. This is because the unskilled jobs and the

unemployed people may not have the ability to pay off their credit card bill as they may have lower income or no incomes.

Baek & Hong (2004) find that employed people were more likely to have credit card debts than the unemployed or people who are not working. Also Zhu & Meeks (1994) find people who were employed full time have larger amounts of credit card outstanding balances compared to those who were not employed. The reason for this is that the employed people have steady income to pay off their credit card bills compared to the unemployed who do not have incomes to pay off credit card bills.

Acceptance Point: Credit card can only be used when there are acceptance points of credit card. The credit card acceptance point referred to the merchants and other shopping places where credit card payments are accepted and the credit cardholders can use their credit cards to make payments (Wong, 2005). For example, with the widely acceptance of credit card by the merchants, consumers can use their credit card to make purchase in any shopping malls, supermarkets and retailers including restaurants, gym and hotels.

Worthington's (2003) study of Chinese payment card market revealed that it is impossible to get a consumer to hold and use credit card if there are not enough acceptance points. Worthington (1998) also found that the increase in the number of

merchants accepting payment by credit card would increase the consumers' usage of credit cards. This is because more places that accept credit card payment will increase the probability that the consumers will use their credit card. Consumers will rely more on credit cards once they get used to the credit card payment and feel the convenience of using credit cards.

2.5 Features of Credit Card

Credit cards provide consumers a secure, reliable and convenient means of payment (Chakravorti, 2003). The use of credit cards gives consumers access to safe and secure transaction without the need to exchange cash. Further, credit card can help users to keep their money secure until they spend. Stolen cash is gone for good, but a stolen credit card can be cancelled with just a phone call, giving the user a safe and effective means to minimize his or her losses in the event of theft and robbery (Autoworld, 2009). Even if a credit card is stolen or lost, the user can report the loss to the bank immediately, and it is easy to track each transaction since it is recorded by the bank. In addition, the banks will likely bear the loss in 48 hours before the user reports the loss (Li, 2011).

The credit card offers consumers with unprecedented convenience as a currency transaction that is accepted worldwide. Worthington, Thompson & Stewart (2011) find that young affluent Chinese cardholder believed credit card is more convenient

than paying with cash, particularly convenient and safety when shopping online and traveling.

The use of credit card can avoid currency exchange rate risk when consumers travel overseas. In addition, the credit card can be used to book hotel, airline ticket online, etc. before the consumers start to travel. Consumers also receive some discount when booking online using their credit cards. If consumers book the hotel room online using their credit card then they will not be penalized if no room is available when they arrive at the hotel.

Credit cards have different credit limit. Consumers can choose different credit limit card to use. For consumers who are married with large household size and higher income will probably have higher daily expenditures (Steifle, 1994; Chien & Devaney, 2001). These people are more likely to request higher credit limit on their credit card, since higher credit limit will enable them to meet their high expenses.

Consumers in China have several ways to avoid annual credit card fee. For example, when the consumers have used their credit card for 3 to 6 times within a year, they do not need to pay the annual credit card fee, or when they spent a certain among of money with their credit card, such as 2,000 RMB, they also do not need to pay for the annual credit card fee (Wenku, 2012).

Furthermore, the credit card can be useful when people do not carry enough cash or not carrying cash, especially when traveling. It is difficult to borrow money overseas, the credit card becomes useful in this situation. Credit card can relieve the consumers' financial stress when they have to make a large amount of one-time payment. They can apply for installment payment to repay the credit card bill after they have use the credit card to pay for the large amount of one-time payment (Xici, 2012).

Many banks credit cards have cooperated with retailers to promote their credit cards, for example, when a consumer spend certain amount of money with one retailer by using his/her credit card, the retailer will gives him/her 20% discount per credit card transaction. Consumers also can get some discounts when they use credit card to buy movie tickets or pay for the dinner in some restaurants. Consumers can earn credit card points when they use their credit card and the points can be used to exchange for gifts or be in the draw for prizes (Xici, 2012).

For some parents with their children studying in other cities, provinces or countries, they can apply for a supplementary card for their children where the parents can basically take control of the children's expenditure (Xici, 2012).

Other advantages of a credit card include offer free accident insurance, medical

insurance, trailer and car wash, bank or airport VIP lounge service, etc. (Xici, 2012). All features and advantages of a credit card can only be achieved if the consumers can get the credit card. Thus, the application process for a credit card is very important. The domestic banks in China have simplified the credit card application process where consumers no longer need to provide many documents to the bank to apply for a credit card. The consumers can even apply for a credit card online. This encourages many consumers to start to apply for and use credit card (Sina, 2011).

2.6 Demographic Factors

Armstrong & Craven (1993) demonstrated that females tend to have a higher number of credit cards than males, but found that on average, females carry lower balances. Hayhoe, Leach, & Turner (1999) found that females are more likely to have four or more credit cards. Further higher income consumers are the main target for the credit card issuers. Studies by Mandell (1972) and Awh & Walters (1974) revealed that family income, education and socio-economic standing were the main determinants for credit card usage. This is further evidenced by credit card companies targeting higher income group consumers. Historical evidence demonstrated that the rising income and the increased in the purchasing power of the household are among the most significant factors during the 1982s and 1990s that resulted in increasing demand for credit card facilities (Kara et al., 1994). According to a research by Mathews & Slocum (1969), people who are in the higher social

class are more likely to use credit cards as a convenient payment tool. Awn & Waters (1974) state that older users are more likely to be inactive user of credit card, and people with higher socioeconomic status are more likely to be active user of credit card.

In China, credit-card companies are targeting young, affluent users who desire a better lifestyle, are more inclined to use bank loans to fund purchases of cars and mortgages. Consumers under the age of 35 with assets of more than 500,000 RMB hold the highest number of cards among all Chinese groups, with 2.6 on average (Leung, 2011). Also 67.4% of Chinese credit card users are male, 32.6% are female, and for the users in different age group, the young people who are between 25 to 29 years old (41.5%) are the major user of credit cards (Yang, 2011).

Chen & Li's (2010) study also found Chinese consumers who are 26 to 35 years old have high knowledge of credit card. However, when the age increases the consumers' knowledge of credit card reduces, since older people may have fewer chances to use the credit card in their spending patterns. Education level has a positive relationship with the knowledge level of credit card. Monthly incomes have a positive relationship with the use of credit card. As income level increases, consumers' use of credit card increases too (Chen & Li, 2010).

People in different demographic groups have different emotion¹ attitude toward credit card. People who are between 26-35 years old have the highest emotion attitude toward credit card, as their age increase, their emotion attitude will decrease. Consumers who are single or married with no children have a stronger positive emotion attitude towards credit card (Chen & Li, 2010). People who have strong positive emotion attitude toward credit card are usually young people. A possible explanation for this is that young people feel the use credit card have become a necessity and think the credit card payment is more convenient than using cash (Yuan, 2013).

2.7 Summary

Chapter Two begins with an overview of development of credit card in general, followed by the credit card market in China and the different factors that influence the consumers' decision in using credit cards. The advantage of credit card and the different demographic group of credit card users were discussed in this chapter as well. The following Chapter details the theoretical and empirical methodologies, sampling and data collection.

_

¹ To be recognized, and willingness to use.

CHAPTER THREE RESEARCH METHODOLOGY

3. Introduction

Chapter Three begins with a description of the research methodology. The empirical framework in this research is derived from the qualitative choice modelling. Logit analysis is chosen because of the binary nature of the dependent variable, and it is used to determine the factors that influence consumers' decision to use credit card. Ordered logit is used to determine whether there are differences in choosing different level of credit limit base on different demographic characteristics. A discussion of the research design, survey questionnaire development and format, sample size and sampling technique concludes the chapter.

3.1 Empirical Framework

The empirical research is developed based on the qualitative choice analysis, which is widely used in describing decision-makers' choices in areas such as banking, transportation, telecommunication and housing (Ben-Akiva & Lerman, 1985; Train, 1986). A qualitative choice situation is defined as one in which a decision-maker faces a choice among a set of alternatives which satisfy the following criteria:

- 1) The number of alternatives in the set is finite;
- 2) The alternatives are mutually exclusive; that is, the person's choosing one alternative in the set necessarily implies that the person does not choose another alternative; and

3) The set of alternatives is exhaustive: that is, all possible alternatives are included, and so the person necessarily chooses on alternative from the set (Varian, 1992).

Any choice or decision is represented by a continuous variable is not considered as a qualitative choice situation (Gao, 2011). A qualitative choice model designates a class of models, such as logit and probit, which attempt to related the probability of making a particular choice to various explanatory factors and calculates the probability that the decision-maker will choose a particular choice or decision from a set of choices or decisions(J_n), given data observed by the researcher. This choice probability (P_{in}) depends on the observed characteristics of alternative i (z_{in}) compared with all other alternatives (z_{jn} , for all j in J_n and $j\neq i$) and on the observed characteristics of the decision-maker (s_n) (Gao, 2011). The choice probability can be specified as a parametric function of the general form:

$$P_{in} = f(z_{in}, z_{in}, s_n, \beta)$$
 (3.1)

Where f is the function relating the observed data to the choice probabilities specified up to some vector of parameters (β) .

By relating qualitative choice models to utility theory, a clear meaning of the choice probability emerges from the derivation of probabilities from utility theory; the utility from each alternative depends on various factors, including the characteristics of both alternative and decision-maker (Gao, 2011).

By labelling the vector of all relevant characteristics of person n as r_n and the vector of all characteristics of alternative i chosen by person n as x_{in} , we can write the utility as a function of these factors,

$$U_{in} = U(xi_n, r_n) \tag{3.2}$$

Different qualitative choice models are obtained by specifying different distributions for unknown component of utility (ϵ_{in}) and deriving functions for the choice probabilities (Train, 1986; Ben-Akiva and Lerman, 1985; Greene, 1990). The logit model is used in this research because of the binary choice of the dependent variable.

The choice probabilities are expressed as follows (Train, 1989; Ben-Akiva and Lerman, 1985; Greene, 1990):

$$P_{\rm in}=1/(1+e-\mu^{\rm [Vin-Vjn]})$$
 (3.3)

3.2 Research Model

The binary logit model is applied to the first and third research objectives where the consumers only face a simple binary choice of use or do not use the credit card. The ordered logit model is used in second objective, where the consumer facing a choice of using credit card or not depend on different level of credit limits. The logit model

attempts to predict the probability of credit card use in urban China related to a set of explanatory factors. The ordered logit model is used to predict the probability of credit card use based on different level of credit limit on the credit card.

In research objective one, with regards to the perspective of making a credit card use decision, the consumers are faced with a simple binary choice situation: have or not have a credit card. The consumer's utility associated with having a credit card is denoted as U_{1n}, and the utility associated with not having a credit card is denoted as U_{0n} , it can be expressed as (Gao, 2011):

$$U_{in} = V_{in} + \varepsilon_{in} \quad \forall \quad i \in J_n \text{ and } J_n = \{0, 1\}$$

$$(3.4)$$

The consumer will choose to have a credit card if $U_{1n} > U_{0n}$, the utility of each choice (Vin) depends on the vector of observable attributes of the choice and the vector of observable consumer characteristics. The error term (ε_{in}) , which includes all unobservable and excludes the consumer characteristics, is assumed to be independently distributed. The choice probability of $U_{1n} > U_{0n}$ is given by $P_{1n}^* = Pr_n$ $(U_{1n}>U_{0n})=1/(1+e^{-\mu (V_{1n}-V_{0n})})$, where $\mu>0$ (Gao, 2011). Hence, the parametric functional form of the logit model can be written as follows:

$$Y_{in}^* = f(CO, CL, IR, AP, AF, RP, AP_2, X_1, X_2, X8, \epsilon)^2$$
 (3.5)

Where

² This is a reduce form eqution

 Y_{in} *= Decision to have a credit card (where 1= have; 0 = do not have)

CO(+) = Convenience

CL(+) = Credit limit

IR (-) = Interest Rate

AP(+) = Acceptant point

AF(-) = Annual fee

RP(+) = Reward program

 $AP_2(+) = Applying credit card process$

Demographic Characteristics:

 $X_1 = Age of consumer (+/-)$

 $X_2 = Male(-)$

 X_3 = Marital status (single) (-)

X₄= Highest educational attainment (-)

 X_5 = Annual household income (+/-)

 X_6 = Occupation (+/-)

 X_7 = Size of household (+)

 X_8 = Duration of work (+)

 $\varepsilon_{in}^* = \text{Error term}$

The explanatory variable in equation 3.5 include convenience, credit limit, interest rate, accept point, annual fee, reward program, apply process, which are all discrete variables, and young age, male, single, education attainment, occupation, annual

household income, size of household, duration of work, which are all dummy variables. Table 3.1.defines the variable used in equation (3.5).

Table 3.1: Variables Definitions used in Equation (3.5)

Variable Name	Description			
Have Credit Card	Dummy variable equal to 1 if consumers have a credit card; 0 otherwise			
Convenience	Discrete variable, use rating scale of 1 to 8 from 1 (least important) to 8			
	(most important)			
Credit Limit	Discrete variable, use rating scale of 1 to 8 from 1 (least important) to 8			
	(most important)			
Interest Rate	Discrete variable, use rating scale of 1 to 8 from 1 (least important) to 8			
	(most important)			
Accept Point	Discrete variable, use rating scale of 1 to 8 from 1 (least important) to 8			
	(most important)			
Annual Fee	Discrete variable, use rating scale of 1 to 8 from 1 (least important) to 8			
	(most important)			
Reward Program	Discrete variable, use rating scale of 1 to 8 from 1 (least important) to 8			
	(most important)			
Apply Process	Discrete variable, from 1 (very easy) to 4 (not easy at all)			
Young age	Dummy variable equal to 1 if the respondent's age is older than 18 and			
	younger than 35 years old; 0 otherwise			
Male	Dummy variable equal to 1 if the respondent is male; 0 otherwise			
Single	Dummy variable equal to 1 if the respondent is single; 0 otherwise			
Bachelor	Dummy variable equal to 1 if the respondent's education level is			
	bachelor; 0 otherwise			
Middle Professional	Dummy variable equal to 1 if the respondent is employed in a middle			
	professional job; 0 otherwise			
Working Duration	Dummy variable equal to 1 if the respondent's duration of employment			
	is 1 to 5 years; 0 otherwise			
Size of household: Three	Number of family members in the household			
Low Income	Dummy variable equal to 1 if monthly income is less than 4000RMB; 0			
	otherwise			
ϵ_{i}	Error term			

Based on the previous studies by Bei (1993), Canner & Cynak (1985), Choi & DeVaney(1995), Steidle (1994) and Wasberg, Hira & Fanslow (1992), there is a

negative relationship between age and being a credit card user. The authors found younger age households are more likely to use the credit card while older age will reduce the likelihood of being an active credit card holder, since older people do not need a significant amount of credit in their spending patterns and have less time to use the credit card than younger people (Awh & Waters, 1974; White, 1975). According to the studies of Armstrong & Craven (1993) and Markovich & DeVaney (1997), they found there is no significant difference in gender and the use of credit card. This is because females and males may use credit card in different ways and shopping for different goods. For example, Kaynak & Harcar (2001) found females are more likely to purchase clothing and household goods and males are more likely to use credit card to purchase electric devices. Also in the previous studies by Kinsey (1981), Steifle (1994), Godwin (1998), Bertaout & Haliassos (2001), White (1975), Kim & DeVaney (2001), Mathews & Slocum (1972), Slocum & Mathews (1970), and Hong & Beak (2004), the coefficient of the borrower characteristics such as marital status, educational attainment, occupation, household income, size of household, duration of work are expected to be positively related to credit card use. However, interest rate is negatively related to credit card use, especially for the lower income consumers, as low income means a higher probability not paying off the credit card bill on time. If the credit card holder misses paying off the credit card bill on time, he or she will have to pay the interest rate incurred. This implies the credit card user pays more than he or she actually spent.

Objective two determines the most important factors that affect the credit card users' choice of different levels of a credit card limit. This is an ordered logit, as the consumers face different level of credit card limit in using credit cards. In an ordinal logistic regression, the event of interest is observing a particular score or in an order. For the consumers choosing different credit limit level on their credit card, the model has following odds:

 θ_1 = prob (credit limit less 1000) / prob (credit limit greater than 1000)

 $\theta_2 = \text{prob } (1001 \text{ to } 5000) / \text{prob } (\text{greater than } 5000)$

 θ_3 = prob (5001 to 10000) / prob (greater than 10000)

 θ_4 = prob (10001 to 20000) / prob (greater than 20000)

 θ_5 = prob (20001 to 50000) / prob (greater than 50000)

 $\theta_6 = \text{prob } (50001 \text{ to } 100000) / \text{prob } (\text{greater than } 100000)$

The last category does not have an odd associated with it since the probability of the last level is 1 including the last odds (Norusis, 2011).

All of the odds are of the form:

$$\theta_{i} = \text{prob}(\text{score} \leq j) / \text{prob}(\text{score} > j)$$
 (3.6)

The equation 3.6 also can be written as:

$$\theta_{j} = \text{prob}(\text{score} \leq j) / (1 - \text{prob}((\text{score} \leq j)),$$
 (3.7)

Since the probability of a score greater than j is 1-probability of a level less than or equal to j (Norusis, 2011).

Hence, the parametric functional forms of the relationship between different credit limit and having a credit card can be written as follows:

$$\ln(\text{prob(event)}/(1 - \text{prob(event)})) = \beta_0 + \beta_1 X_1 + \beta_2 X_2 + \dots + \beta_{13} X_{13}$$
 (3.8)

Where:

X₁=Number of credit card

X₂=How long owning credit card

X₃=Monthly spending

 $X_4 = Annual fee$

 $X_5 = Interest Rate$

 X_6 = Gender of consumer (where 1= male; 0 = female) (+/-)

 $X_7 = Age of consumer (+/-)$

X₈= Marital status (single) (-)

X₉= Highest educational attainment (+)

 X_{10} = Occupation (+)

 X_{11} = Duration of work (+)

 X_{12} = Size of household (+)

 X_{13} = Monthly household income (+)

Table 3.2 defines the variables used in equation (3.8)

Table 3.2 Variables Definitions in used in Equation (3.8)

Variable Name	Description		
Number of credit card	Number of credit card the respondent owns		
How long owning credit	Discrete variable, use 1 to 4 represents different range of time.		
card			
Monthly Spending	Discrete variable, use 1 to 5 represents different range of spending.		
Annual Fee	Discrete variable, use 1 to 4 represents different range of annual fee.		
Interest Rate	Discrete variable, use 1 to 4 represents different range of interest rate.		
Young age	Dummy variable equal to 1 if the respondent's age is older than 18 and		
	younger than 35 years old; 0 otherwise		
Male	Dummy variable equal to 1 if the respondent is male; 0 otherwise		
Single	Dummy variable equal to 1 if the respondent is single; 0 otherwise		
Bachelor	Dummy variable equal to 1 if the respondent's education level is		
	bachelor; 0 otherwise		
Middle Professional	Dummy variable equal to 1 if the respondent is employed in a middle		
	professional job; 0 otherwise		
Working Duration	Dummy variable equal to 1 if the respondent's duration of employment		
	is 1 to 5 years; 0 otherwise		
Size of household: Three	Number of family members in the household		
Low Income	Dummy variable equal to 1 if monthly income is less than 4000RMB; 0		
	otherwise		
ϵ_{i}	Error term		

Previous researches showed that lower income consumers are more likely to have more than one credit card (Slocum & Mathews, 1970). The lower income consumers will use the second credit card to pay off the first credit card bill since they cannot pay off the credit card bill using their income only (Slocum & Mathews, 1970). Monthly spending and the number of credit cards owned are positively related to the credit limit. Abdul-Muhmin & Umar (2007) and Baek & Hong (2004) found consumers who have more than one credit cards tend to have higher debt, where higher debt reflect they have higher credit limit on their credit cards. Further,

according to the finding reported by Bertaout & Haliassos (2001), Chien & Devaney (2001) and Kinsey (1981), household who have higher spending are more likely to have a higher credit limit credit card. Consumers with higher education level will have higher credit limit on their credit card, since higher education level means higher income in the future, and the higher income is likely to increase the demand for consumption and borrowing in the present (Kim & DeVaney, 2001). Higher credit card limit will meet their increasing demand for consumption and borrowings.

Objective three tests the characteristics of the respondents to find out which group of respondents are major credit card users. The logic model is used to test the demographics variables in this objective. The consumer will choose to have a credit card if $U_{1n} > U_{0n}$, the utility of each choice (V_{in}) depends on the vector of observable attributes of the choice and the vector of observable consumer characteristics. The error term (ε_{in}), which includes all unobservable and excludes the consumers' characteristics, is assumed to be independently distributed. The choice probability of $U_{1n} > U_{0n}$ is given by $P^*_{1n} = Pr_n(U_{1n} > U_{0n}) = 1/(1 + e^{--(V_{1n} - V_{0n})})$, where $\mu > 0$. Hence, the parametric functional form of the logit model can be written as below (Gao, 2011):

$$Y_{in}^{**}=f(X_1, X_2, X_3, X_9)+ \varepsilon_{in}^{**}$$
 (3.9)

Where

 Y_{in} **= Decision to have a credit card (where 1= have; 0 = do not have)

-

³ This is a reduce form equation

 $X_1 = Age of consumer (+/-)$

 X_2 = Gender of consumer (male) (-)

X₃= Marital status (single) (-)

X₄= Highest educational attainment (-)

 X_5 = Annual household income (+)

X₆= Occupation (+)

 $X_7 =$ Size of household (+)

 X_8 = Duration of work (+)

 $\mathcal{E}_{in}^* = \text{Error term}$

Table 3.3 defines the variables used in equation (3.9).

Table 3.3: Variables Definitions used in Equation (3.9)

Variable Name	Description		
Have Credit Card	Dummy variable equal to 1 if the respondent has a credit card; 0		
	otherwise		
Young age	Dummy variable equal to 1 if the respondent's age is older than 18 and		
	younger than 35 years old; 0 otherwise		
Male	Dummy variable equal to 1 if the respondent is male; 0 otherwise		
Single	Dummy variable equal to 1 if the respondent is single; 0 otherwise		
Bachelor	Dummy variable equal to 1 if the respondent's education level is		
	bachelor; 0 otherwise		
Middle Professional	Dummy variable equal to 1 if the respondent is employed in a middle		
	professional job; 0 otherwise		
Working Duration	Dummy variable equal to 1 if the respondent's duration of employment		
	is 1 to 5 years; 0 otherwise		
Size of household: Three	Number of family members in the household		
Low Income	Dummy variable equal to 1 if monthly income is less than 4000RMB; 0		
	otherwise		
$\epsilon_{ m i}$	Error term		

3.3 Questionnaire Development and Data

In order to obtain more reliable and accurate results, the survey questionnaire is used and all questions are derived from previous studies. The questionnaire contains four parts: general information of consumer payment behaviours, credit card users, non-credit card users and socio-economic background of the respondents.

The first part of the survey questionnaire comprises of questions which concern with information of the respondents' perception about the payment methods and the attributes of credit card. The questions include how convenience and safety the credit card is, and the features, such as reward program, grace period of the credit card, and the application process of the credit card. The first part of the survey questionnaire also identifies whether the respondent is a credit card user or a non-credit card user.

The second part of the survey questionnaire contains the general credit card information of credit card user, such as the type of credit card, credit card limits, annual fee and interest rate on the credit card. In addition, the questions also include how the respondents use the credit card, frequency of using the credit card and the respondents' behavior in repaying their credit card bill.

The third part of the survey questionnaire investigates the causes of non-credit card

users. The questions measure the factors, which impact the consumers in urban China who decide not to use credit card, such as inconvenient, high annual fee, security concerns, privacy concerns, etc. In addition, the questions also ask the respondents whether or not they want a credit card in the future, and the features a credit card need to have to convince them to obtain a credit card.

Section four of the survey questionnaire contained the socio-economic background of the respondents, which could possibly affect the respondents' ability in obtaining a credit card. All questions are designed to help the researcher to construct the socio and demographic profile of the sample respondents. The general questions in this section include age, gender, education attainment, marital status, occupation, etc.

A pre-test is of the survey questionnaire was conducted to assess the reliability and validity of the survey questions. As the questionnaire is developed specifically for this research, pre-testing helps to clarify the items and questions used in the questionnaire. A total of 30 questionnaires are randomly distributed to both household residents (both credit card user and non-credit card user) in Shijiazhuang aged 18 years and older. The respondents were encouraged to comment on any questions or statements that they thought were ambiguous or unclear. Some minor wording modifications to the questionnaire are made as a result of this process.

3.4 Data

Taking into account the limitation of time and budget and practical difficulties in obtaining the list and information of the targeted population, the research used convenience sampling to select the sampling units. The results from the survey, therefore, cannot be interpreted beyond the sample (Zikmund, Babin, Carr & Griffin, 2010). Cross-sectional data is used to conduct this research. It is useful for comparisons among various population segment (Wong, 1996). This method has the advantage of achieving a good response rate with complete information, good control over the respondent's identity, flexibility during the data collection process, less restriction by the length of the questionnaire, clarity and lack of sequence bias.

Due to the lack of published research in China regarding credit card users, it is necessary to collect primary data to answer the three research objectives of the study. The sample of respondents' responses was drawn from the Shijiazhuang city, Hebei Province of China, from September 2013 to October 2013. A total of 409 completed survey questionnaires were returned. The survey questionnaire comprises the bulk of data required by this research, which determines the factors influencing the respondents' decision to obtain a credit card. The survey data also identifies the borrower characteristics of both credit card user and non-credit card user. Shijiazhuang city is chosen because it is one of the rapidly developing cities in China (Ji, 2010), and it is the capital of Hebei Province and is the economic center

of Hebei Province. As people's living standard improved significantly, more people have access to credit cards.

3.5 Sampling Method

The sample is drawn from household residents (both credit card user and non-credit card user) in Shijiazhuang, the capital city of Hebei Province to examine the factors affecting their decision to obtain a credit card. The data is collected from a convenience sample of individuals, irrespective of their gender, occupation, or income. Convenience sampling is used in the study due to the practical difficulties in obtaining the list and information of our target population. Respondents aged less than 18 years old are excluded from the survey, as it is perceived they might have encountered difficulties interpreting the survey questions. Household residents are approached to participate in the research in front of the residential areas and the shopping malls around Shijiazhuang. We stress clearly "the voluntary participation" criteria before distributing the questionnaire to each participant to fill in.

3.6 Data Collection Method

This study selected both credit card users and non-credit card users as the sample.

The survey questionnaires were distributed in front of four randomly selected residential areas and shopping malls in Shijiazhuang, the capital city of Hebei Province. The survey pack included a copy of the cover letter and the questionnaire.

During the distribution time, the researcher stood around the residential areas and

the shopping malls to ask people to complete the survey questionnaire. A total of 500 questionnaires were distributed during a five-week period. The total final number of useable responses was 409, which yielded a response rate approximately 82 per cent .The non-respondents comprised of all unusable and incomplete survey.

3.7 Sample Size

The study sample size is determined by Cochran's (1963) formula:

$$n_o = \frac{z^2 pq}{e^2}$$

no is sample size

 \boldsymbol{z}^2 is the abscissa of the normal curve that cuts off an area at the tails

e is the desired level of precision

p is the estimated proportion of an attribute that is present in the

q is 1-p

population

Where:

This study chooses the level of confidence at 95% (or \pm 5% precision) and assumes p = 0.5, q = 0.5. Therefore, according to the above formula, total number of sample size will be 385 households.

The total number of survey distributed was 500 to overcome the unusable and

incomplete survey questionnaires and to ensure that 385 or more completed and usable survey are returned. The total final number of useable responses was 409. The response rate was approximately 82% of total number of survey questionnaire distributed and the non-response rate was about 18%. The non-respondents comprised of all refusals, unusable and incomplete survey.

3.8 Summary

Chapter Three discussed the data and methodology used in the research. The questionnaire design, data, sampling method, data collection method and sample response rate were presented as well. Chapter Four discusses the empirical findings, discussion and interpretation of the research findings.

CHAPTER FOUR RESEARCH FINDINGS

4. Introduction

Chapter Four presents the frequencies and statistics that were generated using SPSS (Version 21.0) and Stata (12.0) from the survey sample responses. The empirical results and the findings relating to each objective are also discussed.

4.1 Descriptive Frequencies of the Respondents

A frequency statistics was computed using SPSS (21.0) for the respondents who are credit card users and those who are non-credit card users. Table F.1 shows the descriptive statistics for both credit card user and non-credit card user. From the 409 respondents who have completed the structured survey questionnaires, approximately 65% (267) of the respondents were credit card users while 35% (142) were non-credit card users. The socio-economic characteristics of the respondents were established as follows. The sample respondents comprised of approximately 55.5% (227) females and 44.5% (182) males. The majority of the respondents were in the age group of 26-35 years old (34.5%) and 18-25 years old (19.1%) and most were from the group of working adults in the early years of establishing their careers and possibly married life. The survey results show 70.7% of the respondents was married and 27.9% were single or never married at the time of the survey. The majority of the respondents have either a two-year college degree (42.5%) or a bachelor degree (36.4%). In term of occupation, 42.8% of the respondents worked as

normal company staff, 14.7% engaged in professional jobs and 12.5% of the respondents worked as a company manager. The dominant level of monthly household income was between 5,001 – 7,000 RMB (21.8%) and 4,001- 5,000 RMB (16.4%). From the surveyed respondents, 40.1% of the respondents worked more than 10 years, 25.9% worked between 1 to 5 years and 24.4% worked between 5 to 10 years. The sample statistics also reported that 55.3% of the households comprised of a couple with children, the proportion of adult living alone and immediate and extended family members were 20.0% and 15.6%, respectively. Three people living in the household (including the respondent) with no dependents were considered as the most common current family composition in China.

A comparison of the socio-economic characteristics between credit card users and non-credit card users is presented in Table F.2. Most of the credit card users were female (54.3%) and married (76.8%) at the time of the survey. Similarly, majority of the non-credit card users were also female (57.7%) and married (59.2%) at the time of the survey. In the age category, 40.1% of the credit card users were from the age group of 26-35 years old, 27.3% from the age group of 36-45 years old and 15.4% were between 46-55 years old. For the non-credit card users, 28.9% were from the age group of 18-25 years old, 23.9% were from the age group of 26-35 years old and 18.3% were between 36-45 years old. In terms of education attainment, 44.2% of the credit card users had bachelor degree and 33.7% of the credit card users had Post

graduate, master or PhD degrees. In comparison with credit card users, 41.5% of the non-credit card users had Post graduate, master or PhD degrees, and 39.4% had bachelor degrees. Furthermore, the majority of the credit card users had a relatively higher household income than non-credit card users. In terms of income, 24.7% of the credit card users' income was between 5,001-7,000 RMB, followed by 16.5% with income higher than 10,000 RMB, while majority (17.6%) of non-credit card users had a monthly household income ranged between 4,001-5,000 RMB. With regards to the duration of employment, 41.6% of the credit card users have worked more than 10 years, compared to non-credit card users (37.3%). With regards to the household size, most of the credit card users have a three members living in their families, compared to the non-credit card users whose household size are greater than three members.

4.2 Assessment of the Data

The data was tested in order to verify the statistical assumptions of correlation coefficient of each model and logistic regression analysis have been met. The data comprised of two groups of respondents: credit card users and non-credit card users. There were some questions which were not answered by respondents, so they were treated as missing variables and coded as -9 in the data entry.

4.3 Correlation Coefficients of Explanatory Variables in Each Model

Besides generating the descriptive frequencies of the survey responses, the

correlation matrices were also generated from SPSS (Version 21.0) between the independent variables of the individual models in the research. The Pearson Correlation matrix was used to inspect the correlations between the independent variables. The correlation matrices showed all correlations were well below 0.80 (see Table F.7 and F8), hence, there was no strong correlation between the independent variables in each model.

4.4 Data Level

Due to the dichotomous nature of the dependent variable, such as credit card user versus non-credit card user, the binary logistic regression is used in the research. The socio-demographic characteristics were coded as dummy variables in the analysis.

4.5 Empirical Results and Findings of the Research

The estimates of logic model one and three via maximum likelihood estimates (MLE) have the large sample properties of consistency, efficiency, normality of parameter estimates and validity of the t-test of significance. The ordered logit model was used in model two, since there are more than two categories in the dependent variable, credit limit levels, and each category have a meaningful sequential order (Torres-Reyna, n.d.). The MLE coefficient estimates from the logic analysis have no direct interpretation with respect to the probability of the dependent variable other than indicating a direction of influence on probability. Greene (2000) and Koch (2007)

recommended measuring the change in probability of a particular choice made with respect to a unit change in an independent variable known as the marginal effects.

4.5.1 Results Pertaining to Research Objective One

Research Objective 1: What are the factors that influence consumers' decision to use credit cards?

Research objective one determines the factors that influence the consumers' decisions to use or not to use credit card. Such factors include convenience, interest rate, reward program, technology (acceptance point), credit limit, annual fee, ease in applying credit card, gender, age, marital status, educational level, occupation, working duration, size of household and monthly income. The empirical result and the percentage of the correct predictions are reported in the Table 4.1.

The dependent variable in the model is the consumers' decision to use the credit card and not use the credit card. The result shows approximately 65% of the respondents owned a credit card, while 35% were non-credit card user. The explanatory variables such as convenience and credit limit are measured as interval units. Age, marital status, annual household income, education level, occupation, size of household is dummy-coded variables. Age is divided into three groups: young age (below 36 years old), middle age (36 to 55 years old) and old age (above 55 years old). Similarly, marital status is divided into three groups: single, married and others (divorced or separated and De facto relationship). Monthly household income

includes three groups as well: low monthly income (4,000 RMB or lower), middle monthly income (4,001 RMB to 10,000 RMB) and high monthly income (10,000 RMB and above). Occupation is divided into four groups: professional jobs (lawyer, scientists, engineers, teachers, doctors, etc.), middle professional jobs (civil servant, company managerial staff and owners of private enterprise), normal company staff and others (unemployed and retired). The education level includes four groups such as high school education or lower, two-year College, bachelor degree and others (Post graduate, Master and PhD).

The estimated results in Table 4.1 show that the model fitted the data adequately. The chi squared test strongly rejected the hypothesis of no explanatory power. The percentage of observations that are correctly predicted by the model is 76.5%. At 5% significant level, most explanatory variables were found to be statistically significant to influence the probability of using credit card, except technology, annual fee, male, bachelor, middle profession, one to five working duration and low income.

Table 4.1 Empirical Result (Logit Model of Credit Card Users versus Non-Credit Card Users)

Number of observation= 409

Log likelihood= -197.07672

Wald chi2 (15)= 105.55

Prob > chi2= 0.0000

Degree of freedom= 1

Percentage of Right prediction= 76.5%

Variables	Coefficient	Standard Error	t-statistics	p-value	Marginal Effects
Convenience	0.30927*	0.07071	4.37	0.000	0.06671
Interest rate	-0.30077*	0.06994	-4.30	0.000	-0.06488
Reward Program	0.12775**	0.06362	2.01	0.045	0.02756
Technology (accept	0.07283	0.06174	1.18	0.238	0.01571
point)					
Credit limit	0.10981***	0.64915	1.69	0.091	0.02369
Annual fee	-0.0275922	0.06157	-0.45	0.654	-0.00595
level of difficulty in	-0.7193091*	0.10331	-6.96	0.000	-0.15517
applying credit card					
Male	-0.10557	0.25583	-0.41	0.680	-0.02282
Young age	0.56339***	0.33903	1.66	0.097	0.12194
Single	-0.81130**	0.38674	-2.10	0.036	-0.18384
Education level:	-0.34812	0.08681	-1.34	0.181	-0.07626
Bachelor					
Occupation: middle	0.40042	0.14393	1.39	0.164	-0.04319
professional					
Working duration: 1	0.44119	0.16578	1.33	0.183	0.08403
to 5 years					
Size of household:	0.24722*	0.08929	2.77	0.006	0.05333
Three					
Annual household	-0.24460	0.29699	-0.82	0.410	-0.05355
income: Low income					

^{*}denotes statistical significant at the 0.01 level of significance

As hypothesized, convenience is statistically significant at 1% level in influencing the consumers' decision to use credit card or not to use credit card. The coefficient sign is positive implies that consumers who are concerned about convenience in using credit card are very likely to be a credit card user. The survey result shows

^{**}denotes statistically significant at the 0.05 level of significance

^{***} denotes statistically significant at the 0.1 level of significance

most respondents agreed that credit cards bring them convenience such as traveling or shopping online, and by using credit card, the consumers do not need to carry lots of cash. This is consistent with the findings of Worthington, Thompson & Stewart (2011), where they found young affluent Chinese cardholders believed using a credit card is more convenient than paying with cash. The young affluent Chinese cardholders strongly agreed that credit cards are more useful when travelling and shopping, and it is much easier than paying by cash.

Interest rate is negative and statistically significant at the 1% level. This means the higher the interest rate, the less likely that the consumers will use credit cards. Canner & Luckett (1992) found that the credit card revolvers are more likely to be sensitive to the level of interest rate. This is because when credit card users cannot pay off their credit card bill before due day, they will end up paying higher interest rate to the bank.

Reward program is positive and statistically significant at the 5% level. Reward program means consumers can use the credit card points, where they earned when using credit card to make payment or to exchange for some household goods without spending extra money. The result implies that a good reward program will attract more consumers to apply for credit cards. In addition, most respondents strongly agreed that they can get some extra rewards by using credit card, such as

discount on purchase and gasoline reward. There are many other rewards, such as free accident insurance, medical insurance, trailer and car wash, bank or airport VIP lounge service, etc. (Xici, 2012).

Credit limit is positive and statistically significant at 10% level, which suggests that consumers who are concerned about the credit limit will be very likely to be a credit card user, especially for some people with higher expenditures, where lower credit limit card will be insufficient to meet their expenditures. Eunyoung Beak & Gong-Soog (2004) showed that the credit limit reflects the ability to borrow, consumers who have higher ability to borrow may prefer a higher credit limit card, as the higher credit limit allows them overdraft more money from the credit card.

The level of difficulty in applying a credit card is negative and statistically significant at the 1% level. This means less people will apply for a credit card if the application process is difficult. The easier the applying process the more consumers will apply for the credit card. This is supported by the finding by Sine (2006) where consumers only need to fill up a form in the bank and almost all consumers can get a credit card with different levels of credit limit, and this has attracted many customers to apply for a credit card.

The age of household head is statistically significant at the 10% level, and the

coefficient sign is correct as hypothesized. This suggests that the younger householders are more likely to use credit card than older householders. Base on the survey results, the largest proportion (54%) of credit card users is from the age group of 18 to 35 years old, this is consistent with Awh & Waters (1974)'s finding that older age reduces the likelihood of an individual being an active credit cardholder. This is also consistent with Kim & DeVaney (2001)'s finding, where they found people who are under 37 years old have the highest likelihood of being a revolving credit card user, and people who are older than 37 years old are less likely to be a revolving credit card user.

The marital status coefficient is also significant at the 5% level and has the correct sign. The negative relationship suggests that single respondents are less likely to have a credit card. Kinsey (1981) and Steifle (1994) find that consumers who are married have higher expenditures, which are more likely to spend more in their daily life or even borrowing money, and the credit card is one of the easiest ways to borrow to purchase goods.

The size of the household coefficient is positive and statistically significant at 1% level. In China, most families have three members in the household, two parents with a child. A household with three members implies the daily spending will be higher than other household size, such as a married couple or a single adult living

alone. Thus, it is necessary for them to use credit cards. This is consistent with Chien & Devaney's (2001) study where household with a large household size are more likely to have specific attitudes toward using credit card and having higher outstanding credit card balances. Bertaout & Haliassos (2001) also found, among the households that have a credit card, the households with more children are more likely to revolve credit card debts because more children result in higher expenditure in their daily life.

Other explanatory variables such as technology, annual fee, male, educational level, occupation, working duration and household monthly income are insignificant but with the correct signs in explaining the consumers' decision to use or not to use credit cards.

In summary, the results of the logit analysis suggest the factors such as convenience, interest rate, reward program, credit limit, ease of applying credit card, age, marital status and size of household influence consumers' decision to use credit card (see Table 4.2).

Table 4.2 Result for Research Objective One

Variables	Supported	Not Supported
Convenience	√	
Interest rate	√	
Reward program	√	
Technology (accept point)		√
Credit limit	√	
Annual fee		√
level of difficulty in applying credit card	√	
Male is negative related		√
Young age is positive related	√	
Single is negative related	√	
Educational level is negative related		√
Occupation is positive related		√
Working duration is positive related		√
Size of household is positive related	√	
Lower monthly income is negative		√
related		

Additional analysis of the data was carried out by ranking the means in descending order of importance (see Table 4.3) towards how the credit card attributes are important to the consumers. For example, the mean figures which are close to 8 indicate that the factors are very important; the figures which are close to 4 indicate that the factor are moderately important; and the figures which are close to 1 indicate that the factors are not important at all.

Table 4.3 Mean Values of Credit Card Attributes Measurement (All Respondent)

Attribute Variables	Mean
Convenience	7.05
Credit limit	5.99
Annual fee	5.66
Interest rate	5.42
Technology (accept point)	4.87
Reward program	4.43

Table 4.3 presents the results of the relative importance of each predetermined attitude factor, ranking in descending order of its influence on the decision to use credit card.

According to data in Table 4.5, convenience is the most important attributes which influences the respondents' decision to use credit cards. This is followed by other factors such as credit limit, annual fee, and interest rate. Technology (acceptance point) and reward program factors have marginal influence of the respondents' credit card use decision.

Table 4.4 Characteristics for Not Using Credit Card

Variable		Frequency (no. of Respondents per option)	Percent (%)
	Yes	22	15.5
Inconvenient	No	120	84.5
	Total	142	100.0
Application fee is too high	Yes	12	8.5
	No	130	91.5
	Total	142	100.0
A	Yes	16	11.3
Annual fee is too high	No	126	88.7
	Total	142	100.0
Carrier for in Ass	Yes	14	9.9
Service fee is too high	No	128	90.1
	Total	142	100.0
Risk of	Yes	39	27.5
	No	103	72.5
accumulating	Total	142	100.0
	Yes	10	7.0
Unable to quality	No	132	93.0
	Total	142	100.0
	Yes	41	28.9
Security concerns	No	101	71.1
	Total	142	100.0
	Yes	23	16.2
Privacy concerns	No	119	83.8
	Total	142	100.0
Difficulty record	Yes	4	2.8
Difficulty record keeping	No	138	97.2
	Total	142	100.0
Don't need a	Yes	57	40.1
Don't need a credit card	No	85	59.9
creuit caru	Total	142	100.0

Table 4.4 shows for most of the non-credit card users, the main reason why they choose not to use a credit card is that they do not need a credit card. This is because

they are still used to cash payment. For examples, the survey result shows 61.1% of the respondents still use cash to make payment when shopping. Previous studies show some consumers have higher savings, and their incomes are enough to pay for their daily expenditures, so they do not need credit cards to supplement their daily expenditures (Ma, 2006). A major reason why consumers do not use credit cards is because they may not have the ability to pay off the credit card bill with their low incomes (Wenku, 2008).

In summary, the result of objective one showed apart from the socio-economic factors, convenience, reward program, credit limit and the level of difficulty in applying credit card are the other factors which significantly impact the respondents' decision to use credit cards.

4.5.2 Results Pertaining to Research Objective Two

Research Objective 2: To determine the factors that impact credit card users' choice of different levels of a credit card limit.

Research objective two investigates whether there are differences in choosing different level of credit card limits based on the credit card users' characteristics such as gender, age, marital status, education level, occupation, and duration of employment, household size, monthly household income, and number of credit card, credit card using duration, monthly spending, annual fee, and interest rate.

The dependent variable in the model is the credit limit which has 7 ordinal levels. The explanatory variables such as number of credit card, monthly spending are measured as interval units, and the demographic variables such as female and age is dummy coded. Age is divided into three groups: young age (below 36 years old), middle age (36 to 55 years old) and old age (above 55 years old). Similarly, marital status is divided into three groups: single, married and others (divorced or separated and De facto relationship). Annual household income is estimated by monthly household income times twelve includes low monthly income (4,000 RMB or lower), middle monthly income (4,001 RMB to 10,000 RMB) and high monthly income (10,000 RMB and above). Occupation is divided into four groups: professional jobs (lawyer, scientists, engineers, teachers, doctors, etc.), middle professional jobs (civil servant, company managerial staff and owners of private enterprise), normal company staff and others (unemployed and retired). Education level includes four groups such as high school education or lower, two-year College, bachelor degree and others (Postgraduate, Master and PhD).

The estimated result in Table 4.5 shows the model fitted the data well. The likelihood ratio chi-square of 111.24 with a p-value of 0.0000 exhibit that the model as a whole is statistically significant. The percentage of observations that are correctly predicted by the model is 35.6%. The result of the ordered logit analysis for the credit card users suggests that the number of credit cards, credit card duration,

monthly spending, interest rate, bachelor and normal staff are statistically significant at the 10% level. Table 4.6 shows the marginal effect of each of the credit limit level from outcome 1, the lowest credit limit, to outcome 7, the highest credit limit.

Table 4.5 Empirical Results (Ordered Model of Credit Card Users)

Log likelihood= -406.32058

Number of observations= 267

likelihood ratio chi-square= 111.24 Probability > chi-square= 0.0000

pseudo-R-squared= 0.1204

Percentage of Right prediction= 35.6%

Credit limit	Coefficient	Standard Error	t-statistics	p-value
Number of credit	0.491744*	0.1404191	3.50	0.000
card				
Credit card use	0.4053372*	0.1320661	3.07	0.002
duration				
Monthly spending	0.7872934*	0.1423028	5.53	0.000
Annual fee	0.0751981	0.063485	1.08	0.236
Interest rate	-0.4432509*	0.142405	-3.11	0.002
Female	0.2531265	0.2310961	1.10	0.273
Young	0.3450977	0.2710085	1.27	0.203
Single	-0.2521366	0.3650115	-0.69	0.490
Bachelor	0.465596***	0.24168	1.93	0.054
Normal staff	-0.473073***	0.251116	-1.88	0.060
1 to 5 working	-0.515839	0.31868	-1.62	0.106
duration				
Household size	-0.0162363	0.0836358	-0.19	0.846
(Three)				
Low income	-0.1721339	0.2870242	-0.60	0.549

^{*}denotes statistical significant at the 0.01 level of significance

^{**}denotes statistically significant at the 0.05 level of significance

^{***} denotes statistically significant at the 0.1 level of significance

Table 4.6 Marginal Effect of Each Outcome of Ordered Logit

	Marginal Effect						
Variables	Outcome1	Outcome2	Outcome3	Outcome4	Outcome5	Outcome6	Outcome7
Number of	-0.01902	-0.05596	-0.04582	0.04475	0.05209	0.01957	0.00439
credit card							
Credit card	-0.01567	-0.04613	-0.03777	0.03688	0.04293	0.01613	0.00361
use duration							
Monthly	-0.03044	-0.08961	-0.07335	0.07164	0.08339	0.03133	0.00702
spending							
Annual fee	-0.00291	-0.00856	-0.00701	0.00684	0.00796	0.00299	0.00067
Interest rate	0.01714	0.05045	0.04129	-0.04033	-0.04695	-0.01764	-0.00396
Female	-0.00991	-0.02896	-0.02332	0.02329	0.02665	0.00999	0.00224
Young	-0.01356	-0.03953	-0.03163	0.03178	0.03627	0.01361	0.00305
Single	0.01042	0.02973	0.02221	-0.02504	-0.02576	-0.00945	-0.00211
Bachelor	-0.016906	-0.05082	-0.04488	0.03721	0.05098	0.01991	0.00451
Normal	0.01895	0.05462	0.04256	-0.04432	-0.04926	-0.01843	-0.00413
staff							
1to5 work	0.02265	0.06245	0.04257	-0.05414	-0.05101	-0.01843	-0.00409
duration							
household	0.00063	0.00185	0.00151	-0.00148	-0.00172	-0.00065	-0.00014
size(Three)							
Low	0.00690	0.01998	0.01558	-0.0165	-0.01789	-0.00663	-0.00148
income							

The results in Table 4.5 show the number of credit cards, credit card duration and monthly spending are positive and statistically significant at 1% level and the variable bachelor is positive and statistically significant at 10% level. The variable interest rate and normal staff are negative and statistically significant at 1% and 10% level, respectively. However, Table 4.6 shows outcomes 1, 2 and 3, which represent the lower credit limit level and credit limit range is from less than 1,000 RMB to 10,000 RMB, the factors of number of credit cards, credit card use duration, monthly spending and bachelor exhibit negative signs, while the interest rate and normal staff variables exhibit positive signs. For example in outcome 3, where the

credit limit is from the 5,001 to 10,000 RMB, a unit increase in the variables of number of credit card, credit using duration, monthly spending and bachelor will result in a 4.6%, 3.8%, 7.3% and 0.7% decrease in the marginal probability, respectively, that the credit card users' credit limit will be in the range of 5,000 to 10,000 RMB. In contrast, a one unit increase in interest rate and normal staff will result in a 4% and 1% increase in the marginal probability of the credit card user's credit limit will be in this range of credit limit, respectively.

However, from outcome 4 to 7, the higher credit limit level and the credit limit ranges is from 10,001 RMB to above 100,000 RMB. The number of credit card, credit card use duration, monthly spending, bachelor degree, interest rate and normal staff exhibit the same signs in both Table 4.5 and 4.6. For example in outcome 4, where the credit limit is from 10,001 to 20,000 RMB, a unit increase in the number of credit card, credit use duration, monthly spending and bachelor will result in a 4.4%, 3.7%, 7.2% and 1.4% increase in the marginal probability that the credit card users' credit limit will be in the range of 10,001 to 20,000 RMB, respectively. In contrast, a unit increase in interest rate and normal staff will result in a 4.0% and 1.4% decrease in the marginal probability of the credit card users' credit limit will be in the similar range, respectively.

A unit increase in the number of credit card, credit card use duration and monthly

spending will result a decrease in the probability that the consumers will use a lower credit limit card but increase the probability that consumers will use a higher credit limit card. This means with more credit cards in hands or higher monthly spending and longer credit card use duration, the respondents with higher expenditures, which can be met by higher credit limits. This is similar to Baek & Hong's (2004) study, where they found consumers with a large number of credit cards have higher ability to borrow, which imply they have higher credit card limit. Abdul-Muhmin & Umar (2007) and Baek & Hong (2004) found consumer who own more than one credit cards tend to have higher debt, where higher debt reflect they have higher credit limit on their credit cards. Further according to the finding reported by Bertaout & Haliassos (2001), Chien & Devaney (2001) and Kinsey (1981), household who have higher spending are more likely to have a credit card, and higher credit limit credit card will be needed to meet their higher spending.

A unit increase in the interest rate will result in an increase in the probability that the consumers will hold a lower credit limit card but decreasing the probability that the consumers will hold a higher credit limit card. This is because a higher interest rate implies the credit card user will need to pay more if they cannot pay the credit card bill on time. Further, higher credit limit means consumers can overdraft more money from their credit cards, but will need more money to pay off when the credit card bill is overdue. This is consistent with Canner & Luckett's (1992) study where credit

card revolvers are more likely to be sensitive to the level of interest rate when credit card users cannot pay off their credit card bills before due day. They end up paying higher interest rate to the bank.

A unit increase in the respondent having a bachelor degree will result in a decrease in the probability that the respondent will hold a lower credit limit card but increase the probability he or she will hold a higher credit limit card. Consumers with higher education imply that they have a good job and higher incomes in the future, which will probably increase their consumption. Thus, a higher credit card limit will need to meet their increased consumption. This is similar to Delener & Katzenstein's (1994) result, where they found respondents with a high school degree are more likely to use credit cards. Since higher education level means higher income in the future, and the higher income is likely to increase the demand for consumption and borrowing in the present (Kim & DeVaney, 2001). Higher credit card limit will meet their increasing demand for consumption and borrowings.

In contrast, a unit increase in the normal staff will increase the probability that the consumers will be holding a lower credit limit card but reduces the probability that consumers will be holding a higher credit limit card. This is because a normal staff in China does not command high income, which will reduce their ability to borrow money from credit card, so they may prefer to use a lower credit limit credit card as

they cannot afford the higher credit limit. This is similar to Zhu & Meeks's (1994) study, where they found people who are employed full time had larger amounts of credit card outstanding balances compare to those who were not employed.

Other explanatory variables such as annual fee, female, young, single, one to five years working duration, three people household size and low income are insignificant but have the correct signs.

In summary, the result and findings of the ordered logit analysis suggest there are differences in choosing credit card limits based on the discriminatory factors such as number of credit card, credit card use duration, monthly spending, interest rate and education level, (see Table 4.7).

Table 4.7 Result of Research Objective Two

	Supported	Not
		supported
Number of credit card positive related	\checkmark	
Credit card use duration positive related	\checkmark	
Monthly spending positive related	√	
Annual fee positive related		√
Interest rate negative related	√	
Female positive related		√
Young positive related		√
Single negative related		√
Bachelor positive related	√	
Normal staff negative related	√	
One to five year working duration negative		√
related		
Three people household size negative related		√
Low income negative related		√

4.5.3 Results Pertaining to Research Objective Three

Research Objective 3: Whether different demographic characteristics impact credit card users?

Objective three determines whether different demographic characteristics have impact on the consumers' decisions to use credit card. The demographic characteristics include male, young age, single, bachelor degree, middle professional, one to five years working duration, three people household size and low income, (see Table 4.1).

The estimated coefficients of young age, single and three people household size are statistically significant at the 10% level and have the correct signs, where young age and three people household size are positive related to credit card users while single is negative related. The result shows 65.3% of the respondents are credit card users and 34.7% are non-credit card users. For the credit card users, most of them are in the young age group (18 to 35 years old). The single factor is negatively related to credit card use, where single person is less likely to use a credit card than married people, since they have less daily spending than the married people. The survey result shows amount the credit card users, 76.8% of the respondents are married. Further, the size of household is positive related to the use of credit card, where most families in China have three members, a household with a large family size will probably increase their daily spending, and also will increase their likelihood of

using credit card.

Other explanatory variables such as male, bachelor, middle professional, one to five years working duration and low monthly income are statistically insignificant, but have the correct signs. In summary, the results of the logit analysis suggest the factors such as young age, single and size of household impact consumers' credit card use (see Table 4.8).

Table 4.8 Result for Research Objective Three

Variables	Support	Not support
Male is negative related		√
Young age is positive related	√	
Single is negative related	√	
Educational level is negative related		√
Occupation is positive related		√
Working duration is positive related		√
Size of household is positive related	√	
Lower annual income is negative related		√

4.6 Marginal Effect Analysis

Maddala (1991) and Liao (1994) recommended calculating changes in probabilities to show the magnitude of the marginal effect. This refers to the partial derivatives of the non-linear probability function evaluated at each variable's sample mean (Liao, 1994; Pindyck & Rubinfeld, 1991).

For example, in order to identify the most and least important variables influencing the respondent's decision whether to use a credit card or not to use credit card, the marginal effect for each of the estimated coefficients in the empirical model were calculated. The marginal effect reveals the marginal change in the dependent variable given a unit change in a selected independent variable, holding other variables constant (Liao,1994). The marginal effect indicated the level of importance for the estimated coefficients in the empirical model.

Table 4.9 Marginal Effects of Credit Card Users and Non-Credit Card Users

Factor	Marginal effect	Ranking
Convenience	0.0667168	4
Interest rate	-0.064882	5
Reward program	0.0275585	7
Credit limit	0.0236895	8
level of difficulty in apply credit card	-0.1551708	2
Young	0.1219376	3
Single	-0.1838373	1
Three people household size	0.0533299	6

Table 4.9 shows the marginal effects of the variables of objective one, ranking from the maximum impact to the minimum impact, the single factor has the maximum impact on the consumers' decision to credit card use. Also in the table shows, a unit increase in the importance of convenience and reward program will result in an estimated 7% and 3% increase in the probability of consumers to use credit card, respectively. A unit increase in the importance of interest rate will result in an estimated 6% decrease in the probability of consumers to use credit card. In contract, a unit increase in the importance of credit limit will cause a 2% increase in the probability of consumers to use credit card. The difficult process in apply for a credit card will result in a 16% decrease in the probability of consumers to use credit card.

In terms of the demographic characteristic, if a consumer is in the young age group the probability the consumer will use credit card will increase by 12%. Similarly, a consumer who is single will result an 18% decrease in the probability that he or she will use credit card. In contrast, a household with three people will result in a 5% increase in the probability the household will use a credit card.

Table 4.10 Marginal Effect of Factors Impact Consumers Decision to Use Credit Card

	Marginal Effect							
Variables	Outcome1	Rank	Outcome2	Rank	Outcome3	Rank		
Number of credit card	-0.01902	5	-0.05596	5	-0.04582	5		
Credit card use duration	-0.01567	3	-0.04613	3	-0.03777	3		
Monthly spending	-0.03044	6	-0.08961	6	-0.07335	6		
Interest rate	0.01714	2	0.05045	2	0.04129	2		
Bachelor	-0.01619	4	-0.05082	4	-0.04488	4		
Normal staff	0.01895	1	0.05462	1	0.04256	1		
Variables	Outcome4	Rank	Outcome5	Rank	Outcome6	Rank	Outcome7	Rank
Number of credit card	0.04475	2	0.05209	2	0.01957	3	0.00439	3
Credit card use duration	0.03688	4	0.04293	4	0.01613	4	0.00361	4
Monthly spending	0.07164	1	0.08339	1	0.03133	1	0.00702	1
Interest rate	-0.04033	5	-0.04695	5	-0.01764	5	-0.00396	5
Bachelor	0.03721	3	0.05098	3	0.01991	2	0.00451	2
Normal staff	-0.04432	6	-0.04925	6	-0.01843	6	-0.00413	6

Table 4.10 shows the marginal effect variables in the seven outcomes of the credit limit level, from the credit limit with less than 1,000 RMB to above 100,000 RMB. In the first three outcomes, where the credit limit is from less than 1,000 RMB to

10,000 RMB, which represent lower credit limit level, interest rate has the maximum impact on their decisions in choosing a lower level of credit limit card.

For outcome 4 to outcome 7, where the credit limits are from 10,001 RMB to 100,000 RMB and above, which represent higher credit limit level, monthly spending has the maximum impact on their decisions in choosing a higher level of credit limit card. Thus our results show that "monthly spending" is most important for both high and low credit limits. However, the impact was different. It reduces the probability of having a credit card in the "low credit limits" group, but increases the probability of having a credit card in the "high credit limits" group.

Table 4.11 Marginal Effect of Demographic Factors of Credit Card Users and Non-Credit Card Users

Factors	Marginal Effect	Ranking
Young	0.1219376	2
Single	-0.1838373	1
Three people household size	0.0533299	3

Table 4.11 shows the calculated marginal effect of the significant demographics variables in objective three. If a consumer is single there is an 18% decrease in the probability that he or she will be a credit card user. In contrast, if a consumer is in the young age group there is a 12% increase in the probability he or she to be a credit card user. If a household has three people there is a 5% increase in the probability the household will be a credit card user.

4.7 Summary

Chapter Four discussed the descriptive statistic results generated from the surveyed respondents. The empirical findings of the three models used to answer the three research objectives were discussed. The following chapter presents the conclusions and summaries of the research findings including limitations and recommendations for future research.

CHAPTER FIVE RESEARCH CONCLUSIONS

5. Introduction

This chapter provides a summary of the research, reviews the findings, and provides several conclusions based on the results and discussions presented in Chapter Four.

The research limitations and avenues for future research are also discussed.

5.1 Overview and Summary

The rapid development of credit card in China can be seen in many areas, such as the increasing number of cards issued and the increasing number of acceptance point among the retailers. However, credit card usage in China did not increase significantly; hence the analysis of what has impact Chinese consumers on their decisions of credit card use is important. However, consumers' exhibit different socio-economic characteristics, and different socio-economic characteristics have different impact on the use of credit card. This research investigates the factor that influence consumers' use of credit card in China, and more importantly, this study identifies why Chinese consumers use or do not use credit card.

There are three objectives addressed in the research:

- 1) To identify the factors that influence consumers' decision to use credit cards.
- 2) To determine the factors that impact credit card users' choice of different levels of a credit card limit.

 To examine whether different demographic characteristics impact credit card users versus non-credit card users.

5.2 Conclusions Pertaining to Research Objective One

Research Objective One: To identify the factors that influence consumers' decision to use credit cards.

Research objective one was satisfied as the factors affecting the consumers' decision to use credit card in urban China were identified. The results show that the consumers' are concern about convenience, interest rate, reward program, credit limit and the level of difficulty in applying credit card in their decision to use credit card. The result also shows consumers feel the use of credit card is more convenience, especially travelling overseas and shopping online. The result supports the findings of Worthington, Thompson & Stewart (2011), where they found young affluent Chinese cardholders believed that credit card is more convenient and easier than paying with cash, particularly when shopping online and traveling overseas. In addition the young affluent Chinese are strongly agreed that there are more advantages to paying by credit card than cash, since they accumulate credit card points to redeem gifts in the future.

In terms of demographic characteristic, the results show there is a significant negative relationship between the respondents' decisions in using credit card and the

"Single" marital status. Most of the credit card users are married. The results are similar to the findings of Kinsey (1981) and Steifle (1994) who report that married people are likely to have higher expenditures than non-married people. Thus, it becomes a necessity for the married people to have a credit card to borrow to finance their large living expenses.

There is a significant positive relationship between the respondents' decision in using credit card and the demographic factors such as young age and three people household size. The results are similar to the findings of Awh & Waters (1974), White (1975), Choi & DeVaney(1995), Steidle (1994) and Wasberg, Hira & Fanslow(1992), Bei (1993), Canner & Cynak (1985), Kim & DeVaney (2001), Bertaout & Haliassos (2001). For example, Awh & Waters (1974) also find that older individuals are less likely of being an active credit cardholder. This is because as people get old their spending patterns reduced and they do not need a significant amount of credit thus reduces the use of credit cards. There is a positive relationship between three people household and credit card use. In China, most families have three members in the household that exhibit have higher expenditures than families with less than three people. This is similar to Bertaout & Haliassos's (2001) finding, where they found that consumers with a larger household size will increase their likelihood in using credit card than the consumer living in a smaller household size.

5.3 Conclusions Pertaining to Research Objective Two

Research Objective Two: To determine the factors that impact credit card users' choice of different level of credit card limit.

Research objective two was satisfied, the ordered logit regression result shows that the number of credit card, credit card use duration, monthly spending, and bachelor are statistically significant and positively in relating to the credit card user in choosing different level of credit limit. The factors of interest rate and normal staff are statistically significant and negatively related to the credit card user in choosing different level of credit limit.

A unit increase in the number of credit card, credit card use duration and monthly spending will increase the probability that the credit card user will being holding a higher credit limit card. In contrast, a unit increase in the interest rate will result in a decrease in the probability that the credit card user will be holding a higher credit limit card.

In terms of demographic factors, a unit increase in the bachelor factor will cause an increase in the probability that the credit card user will be holding a higher credit limit card. In contrast, a unit increase in the normal staff factor will cause a decrease in the probability that the credit card user will be holding a higher level of credit

limit card.

5.4 Conclusions Pertaining to Research Objective Three

Research Objective Three: To examine whether different demographic characteristics impact credit card users versus non-credit card users.

Research objective three was satisfied, where young, single and three people household size are statistically significant factors in distinguishing between the credit card users and non-credit card users.

Young respondents are more likely to use credit card, since young consumers see credit card as more convenience to use when making purchase and traveling (Thompson, Worthington & Stewart, 2009). Single people are less likely to use credit cards because their expenditures are lower than married people, so they feel not necessary to have a credit card. In China, most families are three members. A three member household size is the most comment household size in China due to the one child policy. A large household size will have higher daily expenditures, this will increase their likelihood to use credit card as a borrowing instrument to pay their daily expenditures.

5.5 Implications

This research makes a number of contributions to the credit card market and banks in China. First, the research findings provide to the banks with a better understanding of the factors that influence a consumer's decision to use credit cards. With a better understanding of the factors that influence a consumer's decision to use credit cards, banks can strategically promote their credit cards and can improve the usage of credit card. Consumers will consider whether they can afford the credit card or not, so banks should reconcile affordability to borrowers, especially the young and low income borrowers. The results shows the young consumers are more likely to use credit card, but they may have short working duration and low income and savings, this makes the young consumers have less ability to repay the credit card bill. For this reason, banks should reconsider the repayment grace-period and make the credit card payoff more affordable to the young consumers in order to increase the credit card usage. In addition, Chinese consumers should be encouraged to save since savings serve as a step in building their credit worthiness, where they can easily apply for the credit card and the usage of credit card in China will be increased as well.

Secondly, the research also provides banks with a better understanding of credit card users' characteristics that influence their choice of different level of credit limit, since the credit limit is a significant factor in influencing the consumers' decision on

credit card use. Banks can use the background information that consumers provide when applying the credit card to analyse what level of credit limit the consumers qualify. For example, a consumer who has higher monthly spending, have more than one credit card with higher education and higher job level, may prefer a higher level of credit limit. In contract, when consumers show they have less monthly spending and lower job level, they may be more likely to have a lower credit limit card.

Furthermore, convenience and reward program were found to be significant factors in influence consumers' decision on credit card use. In order to attract more consumers to use credit card, the credit card market and banks should make it easier and more convenient for the consumers to use their credit cards. In addition, banks should also improve the credit card reward program system, to make it more rewarding and efficient. This will increase the consumers' likelihood in using credit card. Therefore, banks should cooperate with more merchants, increase the number of merchants that accept credit card payment, including in supermarket and vegetable markets where consumers visit frequently. Such implication supports by the centre government, that it encourages the merchants to accept the credit card payment in order to increase the credit card usage and transactions (KPMG International, 2009). Also in China especially the young consumer group is more likely to enjoy life and spend more money on dining out and entertainment, therefore, the young consumer group can be targeted.

5.6 Limitation and Recommendation for Future Research

The data was collected in Shijiazhuang City, Hebei Province of China. The research was limited by a small sample size, and credit card characteristics vary across different regions and cities. Moreover, the sample respondents could be biased because the surveyed participants were from certain selected residential areas. Furthermore, the sample probability in a different geographic area may appear different in regards to the consumers' characteristics towards the decision to use credit card.

The second limitation of this research is that this study identifies some significant factors that influenced the consumers' decision to use a credit card. However, there may be other factors that could possibly affect consumers' decision to use a credit card, such as lifestyles and locations. For example, location might influence a consumer's decision to use a credit card because if the consumer lives in a village or a town, there is no need for a credit card. Moreover, the consumer may choose not to use a credit card when there is no need to use a credit card in their lifestyle.

This research assesses only the influence of some consumers' characteristics which affect their decision to use credit cards. Future research could address more characteristics or factors such as personal savings, type of credit card and knowledge of credit card. Moreover, a larger sample size should be used, and the respondents

can be selected from different cities across the country, this could effectively increase the accuracy of the research results and findings.

5.7 Conclusions

The objective of this research is to investigate the factors influencing the consumers' decision to credit card use in urban Shijiazhuang. The research findings showed that convenience, reward program, interest rate, credit limit, young age, single and household size factors significantly affect the consumers' credit card use decision in Shijiazhuang. Our survey results show that consumers were more concerned about convenience (such as ease in using credit card) and reward program of credit card when deciding whether to use credit card or not. With regards to demographic characteristics, the consumers who are married and in the age group of 18 to 35 years old are the major credit card users. Most of the families in China have three members and they are more likely to use credit cards.

With regards to credit limit levels among credit card users, our research shows that users of higher credit card limits have higher monthly spending, longer credit card use duration and large number of credit cards with higher education level and higher position level.

References

Ando, A. & Modigliani, F. (1963). The life cycle hypothesis of saving: Aggregate implications and tests. American Economic Review, 53(1), 55-84.

Armstrong, C. & Craven, M. (1993). Credit card use and payment practices among a sample of college students. Proceedings of the 6th Annual Conference of the Association for Financial Counseling and Planning Education. pp. 148-159.

Autoworld. (December 29, 2009). The Power and Convenience of Credit Cards. Retrieved from

http://blog.autoworld.com.my/index.php/2009/12/29/the-power-and-convenience-of-credit-cards/

Awh, R. H. & Waters, D. (1974). A Discriminant Analysis of Economic, Demographic and Attitudinal Characteristics of Bank Charge-Card Holders – A Case Study. The Journal of Finance, 29(June), 973-80.

Baek, E. & Hong, G. (2004). Effects of Family Life-Cycle Stages on Consumer Debts. Journal of Family and Economic Issues, 25(Fall), 359-85.

Baker Mark, Orsmond David. (2010). Household Consumption Trends in China. Personality and Individual Differences, Vol. 5, No. 5, pp. 501-509.

Baker,A.T. and Sckerkaya.A. (1992). 'Globalization of credit card usage: The Case of a developing economy'. International Journal of Bank Marketing. Vol. 10. No. 6. Pp. 27-31

Becker, G. S. (1975). Human capital. 2nd Edition. New York: Columbia University Press for National Bureau of Economic Research.

Bei, L. (1993). Determinants of consumer credit card repayment patterns. Consumer Interests Annual, 39, 147-154

Ben-Akiva, Moshe & Lerman, Steven R. (1985). "Discrete Choice Analysis - Theory and Application To Travel Demand.", MIT Press, Cambridge, Massachusetts, London, England.

Bertaut, C. C., & Haliassos, M. (2001). Debt revolvers for self-control. (Working Paper). Retrieved (March 5, 2002), from http://www.nber.org/_confer/200l/si2001/bertaut.pdf.

BNZ. (n.d.) Asion Banking. Retrieved from

https://www.bnz.co.nz/personal-banking/international-and-migrant/asian-banking

Bryant, W. K. (1990). The economic organization of the household. Cambridge: Cambridge University Press.

Canner, G. B. & Luckett, C.A. (1992). Developments in the pricing of credit card services. Federal Reserve Bulletin, 78, 652-666.

Censky Annalyn. (June 26, 2012). China's middle-class boom. Retrieved from http://money.cnn.com/2012/06/26/news/economy/china-middle-class/index.htm

Chakravorit Sujit. (2003). Theory of Credit Card Networks: A Survey of the Literature. Review of Network Economics, Vol.2, Issue 2.

Chakravorit Sujit. (2006). A Theory of Credit Cards. Ted To.

Chen Nina, Li Wei. (2010). Analysis of the influence of different demographic consumers on the attitude of credit card. Retrieved from http://www.doc88.com/p-947592574184.html

Choi, H. N. & DeVaney, S.A. (1995). Determinants of banks and retail credit card use. Consumer Interests Annual, 41, 148-154.

Chu Kathy. (June 3, 2012). Citigroup gearing up to enter China's credit card market. Retrieved from

http://usatoday30.usatoday.com/money/world/story/2012-03-01/citibank-credit-card s-china/53391046/1

CIW Team Staff. (May 15, 2012). CHARTS: Credit Card Payment in China. Retrieved from http://www.chinainternetwatch.com/1456/credit-card-china/

Credit Card Bible. (Aug 18, 2011). Worldwide Credit Card Statistics. Retrieved from

http://www.creditcardbible.com/credit-card-market/worldwide-credit-card-statistics. html

Duca, J. & Whitesell, W.C. (1995). Credit cards and money demand: A cross -sectional study. Journal of Money, Credit, and Banking, 27(2) 604-623.

Feldman, Barbara J.. (10 July, 2007). Need to Know: Internet credit card safety. Retrieved from

http://www.surfnetkids.com/tech/1076/need-to-know-internet-credit-card-safety/

Gao Xi. (2011). The current development situation and analysis of Chinese credit card industry. Retrieved from

http://www.zhaozhishi.net/licai/xinyongka/index_28978.html

Garcia, G. (1978). Credit cards in an interdisciplinary survey: toward a general theory of consumer behaviour. Research Program in Finance Working.

Garcia, **G.** (1980). Credit card; An Interdisciplinary Survey. Journal of Consumer Research, 4, 327-337.

Godwin, D.D. (1998). Household debt quintiles: Explaining changes 1983-1989. The Journal of Consumer Affairs, 32(2), 369-393.

Greene, William H.,(1990). "Econometric Analysis.", MacMillan Publishing Company.

Haltmaier Jane. (2013). Challenges for the Future of Chinese Economic Growth. International Finance Discussion Papers, No. 1072.

Hayhoe, C., Leach, L., & Turner, P. (1999). Discriminating the number of credit cards held by college students using credit and money attitudes. Journal of Economic Psychology, 20(2), pp. 643-656.

Infographixdirectory. (17 November 2012). Credit card usage in the United States. Retrieved from

http://en.paperblog.com/credit-card-usage-in-the-united-states-355006/

Ji Xue Mei. (2010). "Shijiazhuang Speed" to development and enrich the city spirit. Retrieved from http://www.lunwentianxia.com/product.free.10034901.1/

Kaynak, E. & Harcar, T. (2001). Consumers' attitudes and intentions towards credit card usage in an advanced developing country. Journal of Financial Services Marketing, 6, 24–39.

Kaynak, E., & Harcar, T. (2001). Consumer's attitudes and intentions towards credit card usage in an advanced developing country. Journal of Financial Services Marketing, 6(1), 24.

Khare, A., Khare, A., & Singh, S. (2011). Factors affecting credit card use in India. Asia Pacific Journal of Marketing and Logistics, 24(2), 236-256.

Kim Haejeong & DeVaney A. Sharon. (2001). The Determinants of Outstanding Balances Among Credit Card Revolvers. Financial Counseling and Planning, Volume 12 (1), P67-79.

Kinsey, J. (1981). Determinants of credit card accounts: An application of tobit analysis. Journal of Consumer Research, 8, 172-182.

KPMG International (2009). Card payment in Asia pacific: The state of the nations. Financial services.

Leung. (2011). Consumer lending activity could triple by 2015. Retrieved from http://www.researchinchina.com/Htmls/News/201109/21721.html

Ma Chun Hong. (2006). Why not using credit card? Retrieved from http://www.qingdaonews.com/content/2006-08/26/content_7525669.htm

Mandell, L. (1972). Credit Card Use in the United States. MI: University of Michigan Press, Ann Arbor.

Markovich, Carl A. and DeVaney A. Sharon. (1997). College Senior's Personal Finance Knowledge and Practices. Journal of Family and Consumer Sciences, 89, 3:61-65.

Mathews, H. Lee and Slocum W. John. (1972). ARejoinder to Social Class or Income? Journal of Mar- keting, 36, 1: 69-70.

Mathews, H. Lee, and Slocum W. John Jr. (1969), "Social Class and Commercial Bank Credit Card Usage," Journal of Mar-keting, 33, 71-8.

McKingsey&Company. (2012). Report on Chinese consumption in 2012. Retrieved from http://doc.mbalib.com/view/7ad4a389807bfd350296f3058f06a069.html

Moses Kemibaro. (**Sep 13, 2010**). Equity Bank appointed China UnionPay's sole agent in East Africa. Retrieved from http://www.moseskemibaro.com/2010/09/13/equity-bank-appointed-china-unionpay %E2%80%99s-sole-agent-in-east-africa/

Norusis. (2011). Ordinal Regression. Retrieved from www.norusis.com/pdf/ASPC_v13.pdf

Odinye Ifeoma & Odinye Ifeanyi. (2012). WESTERN INFLUENCE ON CHINESE AND NIGERIAN CULTURES. A new journal of African studies vol. 9

Peng HuiXin, (2012). 2011Blue book of China's credit card industry. Special Report, p34-37.

People's Daily Paper, (21th Nov, 2013). 33-year average annual GDP growth of 9.8%. Retrieved from

http://www.dzwww.com/xinwen/guoneixinwen/201311/t20131121_9223766.htm

Peterson, Richard L. (1976), "The Impact of General Credit Restraint on the Supply of Commercial Bank Consumer Instalment Credit," Journal of Money, Credit and Banking, 8, 527-35.

Sina. (**Aug. 16, 2001**). Why Chinese people do not use credit card too much? Retrieved from http://finance.sina.com.cn/x/20010816/96579.html

Slocum, John W. and Mathews. H. Lee (1970). Social Class and Income as Indicators of Consumer Credit Behavior. Journal of Marketing, 34, 2: 69-73.

Steidle, R. P. (1994). Determinants of bank and retail credit card revolvers: An application using the life-cycle income hypothesis. Consumer Interests Annual, 40, 170-177.

Tan Wei. (2006). Is There Profit in China's Foreign Credit Card Market? Beijing Review Magazine.

The World Bank. (2013). GDP growth (annual %). Retrieved from http://data.worldbank.org/indicator/NY.GDP.MKTP.KD.ZG?page=1

Thompson F. M., Worthington S. (n.d.). 'Future Money' and its impact on the attitudes of young Chinese towards saving

Thompson F.M., Worthington S., Stewart D. (2009). The young affluent Chinese and their attitudes towards using credit cards – a replication study across time and city. ANZMAC.

Train, Kenneth, (1986). "Qualitative Choice Analysis - Theory, Econometrics, and An Application To Automobile Demand.", MIT Press, Cambridge, Massachusetts, London, England.

Union Pay. (2013). General situation about China UnionPay Retrieved from http://corporate.unionpay.com/infoComIntro/infoCompanyIntroduce/zhongguoyinlia ngaikuang/file_3945122.html

Varian, Hal. R.,(1992). "Micro economic Analysis (3rd Edition).", W.W. Norton & Company, 101.

Vilaclara Javier L Santoma, (2009). The credit card industry in China: the rise of a national champion and challenges for the future. China Europe International Business School, No.040

Wasberg, C.A., Hira, T.A. & Fanslow, A.M. (1992). Credit card usage and consumer debt burden of household. Journal of Consumer Studies and Home Economics, 16, 19-32.

Wenku Baidu. (November, 2010). Credit card market plan for Bank of

Communications. Retrieved from http://wenku.baidu.com/view/1af0762a2af90242a995e503.html

White, K. J. (1975). Consumer Choice and Use of Bank Credit Cards: A Model and CrossSection Results. Journal of Consumer Research, 2(June), 10-18.

Worthington Steve, Thompson Mattison Frauke, Stewart B. David. (2011). Credit card in a Chinese cultural context – The young, affluent Chinese as early adopters. Journal of Retailing and Consumer Services, pp. 534-541.

XiCi.net. (2012). Ten advantage of credit card. Retrieved from http://www.xici.net/d160394454.htm

Yang Weiqing. (2011). China Credit Card User Research Report. iResearch Inc. 2011

Yieh, Kaili. (1996). Who Has a Negative Attitude toward Installment Debt in the US. Consumer Interests Annual, 42: 135-40.

Zhou Ming, (2010). 2009 Blue book of China credit card industry. Special Report. P-32-34.

Zhu, Lillian Y. and Meeks ,Carol B.. (1994). Effects of Low Income Families' Ability and Willingness to Use Consumer Credit on Subsequent Outstanding Credit Balances. The Journal of Consumer Aflairs, 28, 2: 403-22.

Zywicki J. Todd. (n.d.). The Economics of Credit Cards. George Mason University School of Law. Law and Economics Working Paper No. 00-22.

Appendix One: Tables

Table F.1: Descriptive Statistics of Socio-Economic Characteristics of All Respondents

Variables		Total Respondents		
		Frequency		
		(no. of	D (0/)	
		Respondents	Percent (%)	
		per option)		
Gender	Male	182	44.5	
	Female	227	55.5	
	Total	409	100.0	
Age	18-25	78	19.1	
	26-35	141	34.5	
	36-45	99	24.2	
	46-55	60	14.7	
	56-65	24	5.9	
	66 above	7	17	
	Total	409	100.0	
Marital status	Single/Never married	114	27.9	
	Married	289	70.7	
	De facto relation	4	1.0	
	Divorced/Separated	2	0.5	
	Total	409	100.0	
Education level	No education	2	0.5	
	Primary school	1	0.2	
	Middle school	7	1.7	
	High school	48	11.7	
	Two-years college	174	42.5	
	Bachelor degree	149	36.4	
	Postgraduate, Master or PHD	28	6.8	
	Others	0	0	
	Total	409	100.0	
Occupation	Professional	60	14.7	
	Self-employer	19	4.6	
	Civil Servant	39	9.5	
	Company Managerial staff	51	12.5	
	Owner of private enterprise	27	6.6	
	Normal company staff	175	42.8	
	Unemployed	11	2.7	
	Retired	21	5.1	
	Others	6	1.5	
	Total	409	100.0	

Variables		Total Res	pondents
		Frequency	
		(no. of	D 4 (0/)
		Respondents	Percent (%)
		per option)	
Duration of	Less than a year	39	9.5
employment			
	1 to 5 years	106	25.9
	5 to 10 years	100	24.4
	Above 10 years	164	40.1
	Total	409	100.0
Composition of	Adult living along	82	20.0
household			
	Couple, with child(ren)	226	55.3
	Couple, with no child(ren)	28	6.8
	Single parent with child(ren)	5	1.2
	Immediate and extended family	64	15.6
	members		
	Others	4	1.0
	Total	409	100.0
Size of household	One person	30	7.3
	Two persons	26	6.4
	Three persons	208	50.9
	Four persons	79	19.3
	Five persons	50	12.2
	Six persons or above	16	3.9
	Total	409	100.0
Household	1000 RMB or less	6	1.5
monthly income			
	1001 to 2000 RMB	38	9.3
	2001 to 3000 RMB	47	11.5
	3001 to 4000 RMB	41	10.0
	4001 to 5000 RMB	67	16.4
	5001 to 7000 RMB	89	21.8
	7001 to 10,000 RMB	59	14.4
	10,000 RMB or above	62	15.2
	Total	409	100.0

Table F.1.1: Profile of Surveyed Respondents

	of Surveyed Respondents		Frequency	
			(No. of	
	Variables		Respondents	Percent (%)
			per option)	
Age	Young age group	Yes	219	53.6
Age	Tourig age group	No	190	46.4
		Total	409	100.0
	Middle age group	Yes	159	38.9
	Wildele age group	No	250	61.1
		Total	409	100.0
	Old age group	Yes	31	7.6
	Old age gloup	No	378	92.4
		Total	409	100.0
Marital Status	Single	Yes	114	27.9
Wartar Status	Single	No	295	72.1
		Total	409	100.0
	Married	Yes	289	70.7
	Wallied	No	120	29.3
		Total	409	100.0
	Others	Yes	6	1.5
	Official	No	403	98.5
		Total	409	100.0
Education level	High school or lower	Yes	58	14.1
		No	351	85.9
		Total	409	100.0
	Two-years college	Yes	174	42.5
	, ,	No	235	57.5
		Total	409	100.0
	Bachelor degree	Yes	149	36.4
		No	260	63.6
		Total	409	100.0
	Others	Yes	28	6.8
		No	381	93.2
		Total	409	100.0
Occupation	Professional	Yes	60	14.7
_		No	349	85.3
		Total	409	100.0
	Middle professional	Yes	136	33.2
	_	No	273	66.8
		Total	409	100.0

	Variables		Frequency (No. of Respondents per option)	Percent (%)
	Normal company staff	Yes	175	42.8
		No	234	57.2
		Total	409	100.0
	Others	Yes	38	9.3
		No	371	90.7
		Total	409	100.0
Duration of employment	Less than a year	Yes	39	9.5
		No	370	90.5
		Total	409	100.0
	1 to 5 years	Yes	106	25.9
		No	303	74.1
		Total	409	100.0
	5 to 10 years	Yes	100	24.4
		No	309	75.6
		Total	409	100.0
	More than 10 years	Yes	164	40.1
		No	245	59.9
		Total	409	100.0
Composition of household	Adult living alone	Yes	82	20.0
		No	327	80.0
		Total	409	100.0
	Couple, with child(ren)	Yes	226	55.3
		No	183	44.7
		Total	409	100.0
	Couple, no child(ren)	Yes	28	6.8
		No	381	93.2
		Total	409	100.0
	Single parent with child(ren)	Yes	5	1.2
		No	404	98.8
		Total	409	100.0
	Immediate and extended family members	Yes	64	15.6
		No	345	84.4
		Total	409	100.0

	Variables		Frequency (No. of Respondents per option)	Percent (%)
	Others	Yes	4	1.0
		No	405	99.0
		Total	409	100.0
Size of household	One person	Yes	30	7.3
		No	379	92.7
		Total	409	100.0
	Two people	Yes	26	6.4
		No	383	93.6
		Total	409	100.0
	Three people	Yes	208	50.9
		No	201	49.1
		Total	409	100.0
	Four and above	Yes	145	35.4
		No	264	64.6
		Total	409	100.0
Monthly household income	Low income range	Yes	132	32.3
		No	277	67.7
		Total	409	100.0
	Middle income range	Yes	215	52.6
		No	194	47.4
		Total	409	100.0
	High income range	Yes	62	15.2
		No	347	84.8
		Total	409	100.0

Table F.2: Descriptive Statistics of the Respondents (Credit card user versus Non-credit card user)

Variables		Credit card	users	Non-credit ca	ard users
		Frequency		Frequency	
		(No. of	Percent	(No. of	Percent
		Respondents	(%)	Respondents	(%)
		per option)		per option)	
Gender	Male	122	45.7	60	42.3
	Female	145	54.3	82	57.7
	Total	267	100.0	142	100.0
Age	18-25	37	13.9	41	28.9
	26-35	107	40.1	34	23.9
	36-45	73	27.3	26	18.3
	46-55	41	39.2	19	13.4
	56-65	9	3.4	15	10.6
	66 above	0	0.0	7	4.9
	Total	267	100.0	142	100.0
Marital status	Single/Never married	59	22.1	55	38.7
	Married	205	76.8	84	59.2
	De facto relation	2	0.7	2	1.4
	Divorced/Separated	1	0.4	1	0.7
	Total	267	100.0	142	100.0
Education level	No education	1	1.3	1	0.7
	Primary school	0	0.0	1	0.7
	Middle school	5	1.9	2	1.4
	High school	30	11.2	18	12.7
	Two-years college	118	44.2	56	39.4
	Bachelor degree	90	33.7	59	41.5
	Postgraduate, Master or PHD	23	8.6	5	3.5
	Others	0	0.0	0	0.0
	Total	267	100.0	142	100.0
Occupation	Professional	39	14.6	21	14.8
	Self-employer	12	4.5	7	4.9
	Civil Servant	29	10.9	10	7.0
	Company Managerial staff	41	15.4	10	7.0
	Owner of private enterprise	20	7.5	7	4.9
	Normal company staff	116	43.4	59	41.5
	Unemployed	4	1.5	7	4.9
	Retired	3	1.1	18	12.7
	Others	3	1.1	3	2.1
	Total	267	100.0	142	100.0

Variables		Credit card	users	Non-credit ca	ard users
		Frequency		Frequency	
		(No. of	Percent	(No. of	Percent
		Respondents	(%)	Respondents	(%)
		per option)		per option)	
Duration of	Less than a year	12	4.5	27	10.0
employment		12	4.5	27	19.0
	1 to 5 years	66	24.7	40	28.2
	5 to 10 years	78	29.2	22	15.5
	Above 10 years	111	41.6	53	37.3
	Total	267	100.0	142	100.0
Composition of household	Adult living along	47	17.6	35	24.6
	Couple, with child(ren)	166	62.2	60	42.3
	Couple, with no child(ren)	16	6.0	12	8.5
	Single parent with child(ren)	3	1.1	2	1.4
	Immediate and extended family members	34	12.7	30	21.1
	Others	1	0.4	3	2.1
	Total	267	100.0	142	100.0
Size of household	One person	22	8.2	8	5.6
	Two persons	14	5.2	12	9.0
	Three persons	157	58.8	51	35.9
	Four persons	42	15.7	37	26.1
	Five persons	24	9.0	26	18.3
	Six persons or above	8	3.0	8	5.6
	Total	267	100.0	142	100.0
Household monthly income	1000 RMB or less	5	1.9	1	2.1
·	1001 to 2000 RMB	17	6.4	21	14.8
	2001 to 3000 RMB	24	9.0	23	16.2
	3001 to 4000 RMB	28	10.5	13	9.2
	4001 to 5000 RMB	42	15.7	25	17.6
	5001 to 7000 RMB	66	24.7	23	16.2
	7001 to 10,000 RMB	41	15.4	18	12.7
	10,000 RMB or above	44	16.5	18	12.7
	Total	267	100.0	142	100.0

Table F.3: Descriptive Statistic of the Respondents' Socio-Economic Factors

Variables	N	Minimum	Maximum	Mean	SD	Variance
Gender	409	0	1	0.56	0.498	0.248
Age	409	1	6	2.59	1.216	1.478
Marital Status	409	1	4	1.74	0.491	0.241
Education level	409	1	7	5.32	0.907	0.822
Occupation	409	1	9	4.65	2.096	4.394
Duration of employment	409	1	4	2.95	1.021	1.042
Composition of household	409	1	6	2.40	1.314	1.726
Size of household	409	1	12	3.37	1.230	1.513
Monthly household income	409	1	8	5.29	1.927	3.713

Table F.4: Statements Pertaining to the Respondents' Attitude toward Credit Cards

			Frequency	Percent (%)
	Valid	Strongly disagree	58	14.2
		Disagree	13	3.2
		Little disagree	32	7.8
I can withdraw cash from my		Neutral	127	31.1
credit card at any ATM		Little agree	28	6.8
		Agree	44	10.8
		Strongly agree	107	26.2
		Total	409	100.0
	Valid	Strongly disagree	41	10.0
Using credit cards, I can		Disagree	12	2.9
make a purchase in any part		Little disagree	22	5.4
of the world without		Neutral	121	29.6
worrying about currency		Little agree	37	9.0
exchange rates or foreign		Agree	36	8.8
currency notes		Strongly agree	140	34.2
		Total	409	100.0
	Valid	Strongly disagree	21	5.1
		Disagree	10	2.4
Condit and offers on officient		Little disagree	31	7.6
Credit card offers an efficient		Neutral	93	22.7
payment solution especially for online purchases		Little agree	47	11.5
for offiffic purchases		Agree	59	14.4
		Strongly agree	148	36.2
		Total	409	100.0

			Frequency	Percent (%)
	Valid	Strongly disagree	20	4.9
		Disagree	15	3.7
		Little disagree	21	5.1
Using a credit card can help		Neutral	108	26.4
me track me expenses and		Little agree	55	13.4
avoid fraud		Agree	69	16.9
		Strongly agree	121	29.6
		Total	409	100.0
	Valid	Strongly disagree	12	2.9
		Disagree	8	2.0
Using credit cards can make		Little disagree	20	4.9
my purchases easier, such as		Neutral	80	19.6
booking a trip, shopping		Little agree	42	10.3
online, or over the phone		Agree	71	17.4
		Strongly agree	176	43.0
		Total	409	100.0
	Valid	Strongly disagree	16	3.9
		Disagree	5	1.2
To '		Little disagree	18	4.4
It is necessary a have a credit		Neutral	92	22.5
card with you when you travel overseas		Little agree	44	10.8
traver overseas		Agree	57	13.9
		Strongly agree	177	43.3
		Total	409	100.0
	Valid	Strongly disagree	15	3.7
		Disagree	7	1.7
Most credit cards off 24-hour		Little disagree	29	7.1
service for lost or stolen		Neutral	81	19.8
cards		Little agree	52	12.7
cards		Agree	64	15.6
		Strongly agree	161	39.4
		Total	409	100.0
	Valid	Strongly disagree	21	5.1
		Disagree	20	4.9
Credit card PIN number		Little disagree	25	6.1
ensure that only I have		Neutral	93	22.7
access		Little agree	54	13.2
access		Agree	53	13.0
		Strongly agree	143	35.0
		Total	409	100.0

			Frequency	Percent (%)
	Valid	Strongly disagree	32	7.8
		Disagree	15	3.7
Credit card cannot be used to		Little disagree	31	7.6
make purchases at any stores		Neutral	73	17.8
without my signature and/or		Little agree	37	9.0
PIN number		Agree	63	15.4
		Strongly agree	158	38.6
		Total	409	100.0
	Valid	Strongly disagree	14	3.4
		Disagree	6	1.5
		Little disagree	18	4.4
I don't need to carry a lot of		Neutral	75	18.3
cash if I have a credit card		Little agree	48	11.7
		Agree	51	12.5
		Strongly agree	197	48.2
		Total	409	100.0
	Valid	Strongly disagree	26	6.4
		Disagree	7	1.7
T 1 . 1		Little disagree	25	6.1
I do not have to worry about		Neutral	80	19.6
my money being snatched by thieves		Little agree	51	12.5
tnieves		Agree	43	10.5
		Strongly agree	177	43.3
		Total	409	100.0
	Valid	Strongly disagree	16	3.9
		Disagree	8	2.0
Condit and offer consumer		Little disagree	26	6.4
Credit cards offer consumer		Neutral	97	23.7
24 hours protections that cash doesn't		Little agree	43	10.5
cash doesn't		Agree	63	15.4
		Strongly agree	156	38.1
		Total	409	100.0
	Valid	Strongly disagree	17	4.2
Credit cards often come with		Disagree	18	4.4
special offers, such as cash		Little disagree	29	7.1
rebates, airline miles for		Neutral	100	24.4
frequent fliers and "points"		Little agree	62	15.2
systems that give free		Agree	63	15.4
products after a certain amount is spend		Strongly agree	120	29.3
amount is spend		Total	409	100.0

			Frequency	Percent (%)
	Valid	Strongly disagree	16	3.9
		Disagree	16	3.9
Credit cards rewards include		Little disagree	24	5.9
cash back rewards, discounts		Neutral	98	24.0
at various merchants, and		Little agree	61	14.9
gasoline rewards		Agree	68	16.6
		Strongly agree	126	30.8
		Total	409	100.0
	Valid	Strongly disagree	15	3.7
		Disagree	11	2.7
C. 1' 1 - ff 11 1-1 - 1 - f		Little disagree	24	5.9
Credit cards offer all kinds of		Neutral	94	23.0
extras/rewards that cash doesn't		Little agree	59	14.4
doesii t		Agree	68	16.6
		Strongly agree	138	33.7
		Total	409	100.0
	Valid	Strongly disagree	14	3.4
		Disagree	8	2.0
C 1'4 1 CC 1'		Little disagree	25	6.1
Credit card offers discounts		Neutral	83	20.3
on purchases with retail		Little agree	65	15.9
partners		Agree	73	17.8
		Strongly agree	141	34.5
		Total	409	100.0
	Valid	Strongly disagree	15	3.7
M		Disagree	16	3.9
Most credit cards offer some form of cash back or other		Little disagree	29	7.1
		Neutral	115	28.1
loyalty rewards to encourage you to use their credit card		Little agree	44	10.8
over a competitor's card		Agree	55	13.4
over a competitor's card		Strongly agree	135	33.0
		Total	409	100.0
	Valid	Strongly disagree	21	5.1
Most cards offer a free travel		Disagree	14	3.4
assistance program such as		Little disagree	24	5.9
travel accident insurance,		Neutral	121	29.6
purchase protection, and car		Little agree	46	11.2
rental loss or damage		Agree	56	13.7
protection		Strongly agree	127	31.1
		Total	409	100.0

Table F.5: Important Factor for Respondent to own a Credit Card with 1 (least important) to 8 (most important)

			Frequency	Percent (%)
Convenience	Valid	1	4	1.0
		2	3	0.7
		3	5	1.2
		4	22	5.4
		5	18	4.4
		6	41	10.0
		7	94	23.0
		8	222	54.3
		Total	409	100.0
Trust (safety)	Valid	1	2	0.5
		2	2	0.5
		3	8	2.0
		4	21	5.1
		5	24	5.9
		6	58	14.2
		7	47	11.5
		8	247	60.4
		Total	409	100.0
Interest rate	Valid	1	39	9.5
		2	25	6.1
		3	26	6.4
		4	47	11.5
		5	41	10.0
		6	65	15.9
		7	63	15.4
		8	103	25.2
		Total	409	100.0
Reward program	Valid	1	79	19.3
		2	35	8.6
		3	27	6.6
		4	69	16.9
		5	47	11.5
		6	53	13.0
		7	40	9.8
		8	59	14.4
		Total	409	100.0

			Frequency	Percent (%)
Technology	Valid	1	58	14.2
		2	36	8.8
		3	27	6.6
		4	60	14.7
		5	55	13.4
		6	36	8.8
		7	47	11.5
		8	90	22.0
		Total	409	100.0
Credit limit	Valid	1	17	4.2
		2	13	3.2
		3	27	6.6
		4	42	10.3
		5	60	14.7
		6	41	10.0
		7	61	14.9
		8	148	36.2
		Total	409	100.0
Annual fee	Valid	1	37	9.0
		2	26	6.4
		3	21	5.1
		4	40	9.8
		5	40	9.8
		6	48	11.7
		7	60	14.7
		8	137	33.5
		Total	409	100.0
Grace payment period	Valid	1	23	5.6
		2	18	4.4
		3	16	3.9
		4	29	7.1
		5	32	7.8
		6	49	12.0
		7	57	13.9
		8	185	45.2
		Total	409	100.0

Table F.6: Importance of the Loss of Financial Control When Choosing Credit Card

			Frequency	Percent (%)
	Valid	Very Unimportant	44	10.8
Using cash		Unimportant	35	8.6
would help		Neutral	134	32.8
avoid getting		Important	63	15.4
into debt		Very important	133	32.5
		Total	409	100.0
When I have	Valid	Very Unimportant	36	8.8
cash I know		Unimportant	32	7.8
how much I		Neutral	119	29.1
have before I go		Important	73	17.8
shopping and		Very important	149	36.4
how much I		Total	409	100.0
have afterwards			409	100.0
Using cash I can	Valid	Very Unimportant	84	20.5
see what I have		Unimportant	37	9.0
spent but with a		Neutral	149	36.4
credit card I		Important	62	15.2
don't know		Very important	77	18.8
don t know		Total	409	100.0
I f1 . 1 f	Valid	Very Unimportant	79	19.3
I feel a loss of		Unimportant	47	11.5
control over		Neutral	134	32.8
expenditure		Important	66	16.1
when using credit cards		Very important	83	20.3
credit cards		Total	409	100.0
XX / . 1 . 1	Valid	Very Unimportant	74	18.1
With credit		Unimportant	59	14.4
cards, I spend		Neutral	115	28.1
too much and didn't save		Important	66	16.1
		Very important	95	23.2
anything		Total	409	100.0
	Valid	Very Unimportant	37	9.0
I have a credit		Unimportant	23	5.6
card and I never		Neutral	100	24.4
owe the bank		Important	65	15.9
money		Very important	184	45.0
		Total	409	100.0

Table F.7: Correlation Coefficient of Objective One and Three

		Convenience	Interest rate	Reward	Technology	Credit limit	Annual fee	Applying process	Male	Young age	Single	Bachelor	Middle professional	One to five year working duration	Size of household: Three	Lower
Convenience	Pearson Correlation	1.000														
Interest rate	Pearson Correlation	0.179	1.000													
Reward program	Pearson Correlation	0.061	0.369	1.000												
Technology	Pearson Correlation	0.131	0.350	0.437	1.000											
Credit limit	Pearson Correlation	0.231	0.123	0.128	0.202	1.000										
Annual fee	Pearson Correlation	0.081	0.306	0.266	0.247	0.320	1.000									
Applying process	Pearson Correlation	-0.189	-0.061	-0.001	-0.064	-0.129	0.036	1.000								
Male	Pearson Correlation	0.010	-0.016	-0.026	-0.064	0.005	-0.096	-0.083	1.000							
Young age	Pearson Correlation	-0.002	0.102	0.241	0.200	0.070	0.103	-0.012	-0.113	1.000						
Single	Pearson Correlation	-0.009	-0.003	0.061	0.100	-0.012	0.010	0.139	-0.096	0.546	1.000					

Bachelor	Pearson	-0.004	-0.028	-0.041	-0.040	-0.054	-0.040	-0.009	-0.034	0.114	0.130	1.000				
	Correlation															
Middle	Pearson	0.034	-0.003	-0.099	-0.061	0.114	0.063	-0.178	0.161	-0.248	-0.218	-0.038	1.000			
professional	Correlation															
One to five	Pearson	-0.035	0.058	0.101	0.018	-0.015	0.063	0.061	-0.092	0.428	0.441	0.039	-0.145	1.000		
working	Correlation															
duration																
Size of	Pearson	0.006	0.054	-0.073	-0.047	0.088	0.000	-0.203	0.073	-0.151	-0.348	-0.038	0.123	-0.211	1.000	
household:	Correlation															
three																
Lower	Pearson	-0.049	0.009	0.136	0.155	0.017	0.014	0.113	-0.092	0.266	0.411	0.043	-0.221	0.248	-0.232	1.000
income	Correlation															

Table F.8: Correlation Coefficient of Objective Two

		No. of credit card	Credit card using duration	Monthly spending	Annual fee	Interest rate	Female	Young	Single	Bachelor	Normal staff	One to five working duration	Size of household: Three	Lower
No. of credit card	Pearson Correlation	1.000												
Credit card using duration	Pearson Correlation	0.244	1.000											
Monthly spending	Pearson Correlation	0.278	0.196	1.000										
Annual fee	Pearson Correlation	-0.0421	-0.007	-0.011	1.000									
Interest rate	Pearson Correlation	0.134	0.044	-0.024	0.047	1.000								
Female	Pearson Correlation	0.033	-0.077	-0.0416	0.021	-0.051	1.000							
Young	Pearson Correlation	-0.043	-0.215	-0.189	-0.117	0.030	0.117	1.000						
Single	Pearson Correlation	-0.104	-0.294	-0.184	-0.099	0.067	0.072	0.474	1.000					
Bachelor	Pearson Correlation	-0.043	-0.022	0.088	-0.082	-0.127	0.002	0.103	0.098	1.000				

Normal staff	Pearson	-0.057	-0.147	-0.203	-0.029	0.003	0.182	0.309	0.207	0.030	1.000			
Normal stan	Correlation													
One to five year working	Pearson	-0.089	-0.320	-0.093	-0.079	0.059	0.125	0.390	0.511	0.032	0.216	1.000		
duration	Correlation													
G: 61 1.11 TI	Pearson	-0.001	0.182	0.003	-0.008	0.008	-0.111	-0.132	-0.398	-0.063	-0.172	-0.261	1.000	
Size of household: Three	Correlation													
	Pearson	-0.091	-0.368	-0.080	0.019	-0.017	0.131	0.253	0.416	0.089	0.251	0.285	-0.246	1.000
Lower income	Correlation													

Appendix Two: SURVEY QUESTIONNAIRE COVER LETTER

To Whom It May Concern

Dear Sir/Madam:

If you are aged 18 years or above, you are invited to participate in a survey that constitutes part of my Master of Commerce and Management thesis at Lincoln University, New Zealand. This is a part of my research project titled "An Empirical Investigation of Credit Card Users in China". The purpose of this research is to investigate the factors influencing consumers' use of credit card in China.

This research is completely voluntary in nature and you are free to decide not to participate at any time during the process of completing the questionnaire and without prejudice, including withdrawal of any information you have provided. However, if you complete the questionnaire and return it to me, it will be understood that you are 18 years of age or older and have consented to participate in this survey and consent to publication of the results of this research with the understanding the anonymity will be preserved.

Your participation is of great assistance to this research. This survey will take maximum 25 minutes to complete. I would be grateful if you would complete the questionnaire and return it to me once you have finished.

Complete anonymity is assured in this survey, as the questionnaire is anonymous. No questions are asked which would identify you as an individual. All responses will be aggregated for analysis only, and no personal details will be reported in the thesis or any resulting publications.

If you have any question about this survey, feel free to contact me on 0221776099 or by email at dongwk1112@live.com. You can also contact my supervisors Dr. Christopher Gan and Dr. Baiding Hu. Dr. Christopher Gan can be contacted at (03) 4230227 or Christopher.Gan@lincoln.ac.nz; and Dr. Baiding Hu can be contacted at (03) 4230231 or Baiding.Hu@lincoln.ac.nz.

Thank you for your kind co-operation and assistance.

Yours sincerely,

Weikang Dong

Master student of Commerce and Management

Research Supervisors:

Dr. Christopher Gan

Professor

Senior Lecturer

Faculty of Commerce

Lincoln University

Dr. Baiding Hu

Senior Lecturer

Faculty of Commerce

Lincoln University

This project has been reviewed and approved by the Lincoln University Human Ethics Committee.

Code	No	
Couc	110.	

Household Credit Card

otherw. measur	ise, es c	please follow the instructions of	given i will be	to a	rovided, please tick your answer answer the questions. Only summ eported. Your participation is volunt	ary
		Section 1. General Info	ormati	ion a	about credit cards	
1.	tha		iods, v		supermarket, retail store, or online ch payment method do you most	
	b.	Check	[]		
	С.	Credit card	ſ	,]		
	d.	Other (please specify)			_ []	
2.	im a. b. c. d. e. f.	Security Control over payment timing Cost Track spending Acceptance for Payment Ease of Use Speed	which	pay [[[[[
3.	imph. i.Co	portant to you when you decide we security control over payment timing ost			following characteristics is the LEAS yment methods to use.	Т
	n.	Speed		[. 1	

4. Below is a series of statements pertaining to your attitude toward credit cards. Please CIRCLE how strongly you agree or disagree with each of the following statements on a scale of 7 to 1. **1-you strongly disagree (SD), 7-you strongly agree (SA).**

, 8,	SD Neutral						
1. I can withdraw cash from my credit	1	2	3	4	5	6	7
card at any ATM							
2. Using credit cards, I can make a	1	2	3	4	5	6	7
purchase in any part of the world							
without worrying about currency							
exchange rates or foreign currency notes							
3. Credit card offers an efficient payment	1	2	3	4	5	6	7
solution especially for online purchases							
4. Using a credit card can help me track	1	2	3	4	5	6	7
me expenses and avoid fraud							
5. Using credit cards can make my	1	2	3	4	5	6	7
purchases easier, such as booking a trip,							
shopping online, or over the phone							
6. It is necessary a have a credit card	1	2	3	4	5	6	7
with you when you travel overseas							
7. Most credit cards off 24-hour service	1	2	3	4	5	6	7
for lost or stolen cards							
8. Credit card PIN number ensure that	1	2	3	4	5	6	7
only I have access							
9. Credit card cannot be used to make	1	2	3	4	5	6	7
purchases at any stores without my							
signature and/or PIN number							
10. I don't need to carry a lot of cash if I	1	2	3	4	5	6	7
have a credit card							
11. I do not have to worry about my	1	2	3	4	5	6	7
money being snatched by thieves							
12. Credit cards offer consumer 24 hours	1	2	3	4	5	6	7
protections that cash doesn't							
13. Credit cards often come with special	1	2	3	4	5	6	7
offers, such as cash rebates, airline miles							
for frequent fliers and "points" systems							
that give free products after a certain							
amount is spend							
14. Credit cards rewards include cash	1	2	3	4	5	6	7
back rewards, discounts at various							
merchants, and gasoline rewards							
15. Credit cards offer all kinds of	1	2	3	4	5	6	7
extras/rewards that cash doesn't							

16. Credit card offers discounts on	1	2	3	4	5	6	7
purchases with retail partners							
17. Most credit cards offer some form of	1	2	3	4	5	6	7
cash back or other loyalty rewards to							
encourage you to use their credit card							
over a competitor's card							
18. Most cards offer a free travel	1	2	3	4	5	6	7
assistance program such as travel							
accident insurance, purchase protection,							
and car rental loss or damage protection							

5. Below is a series of credit card attributes. Please rank them on how important they are to your decision to own a credit card with 1 (least important) to 8 (most important).

a.	Convenience	
b.	Trust (safety)	
c.	Interest rate	
d.	Reward program	
e.	Technology	
f.	Credit limit	
g.	Annual fee	
h.	Grace payment period (2	20 – 25 davs)

6. How important is the **loss of financial control** to you when choosing a credit card? Please circle the number which most accurately reflects how important or unimportant each factor on a scale of 1 to 5, where 1 means "Very unimportant" and 5 means "Very important".

	Very Unimportant		Neutral		Very Important
Using cash would help avoid getting into debt	1	2	3	4	5
2. When I have cash I know how much I have before I go shopping and how much I have afterwards	1	2	3	4	5
3. Using cash I can see what I have spent but with a credit card I don't know	1	2	3	4	5
4. I feel a loss of control over expenditure when using credit cards	1	2	3	4	5
5. With credit cards, I spend too much and didn't save anything	1	2	3	4	5
6. I have a credit card and I never owe the bank money	1	2	3	4	5

7.	How ea	isy is it to apply for a credit (card?				
	a.	Very easy]]			
	b.	Fairly easy]]			
	c.	Somewhat easy	[]			
	d.	Not easy at all	[]			
	e.	Do not know	[]			
8.	Do vou	own a credit card (e.g., Uni	onpav. Vis	a. Mas	terCard. et	:c.)?	
		Yes [] b. No	[]	.,		,.	
	IT YE	ES please go to SECTION 2 a	nd 4; If NC	pleas	e go to SEC	TION 3 and	14
		Section 2	. Credit	Card H	lolders		
1.	Hov	v many credit cards do you	have?		(cards)	
2.	Who	ere did you obtain your cred	dit card? (0	Check a	all that app	lv)	
	a.	Bank	[]	~ PP	,,	
	b.	Work unit	[1			
	c.	Shopping center	ĺ	1			
	d.	Supermarkets	,]			
	e.	Other (please specify)	_	_	[]		
3.	Whi	ich of the following credit ca	ırds do voi	u have	? (Check al	that apply)
	a.	Visa	[1	(,
	b.	Mastercard		, 1			
	C.	Unionpay	ĺ	1			
	d.	America Express	ĺ	1			
	e.	Visa Electron	ĺ	1			
	f.	Other (please specify)			[]		
4.	Whi	ich card do you use the mos	t often (pr	imary	card)?		
	a.	Visa	[]	-		
	b.	Mastercard	[]			
	c.	Unionpay	[]			
	d.	America Express	[]			
	e.	Visa Electron	[]			
	f.	Other (please specify)			[]		
5.	W/hat id	s the current credit line (lim	it) on vour	prima	rv card?		
٥.	a.	RMB less than 1,000	ic, on your	[]		
	۵.			L	,		

b.	RMB 1,001 to 5,000				[]
c.	RMB 5,001 to 10,000				[]
d.	RMB 10,001 to 20,000				[]
e.	RMB 20,001 to 50,000				[]
f.	RMB 50,001 to 100,000)			[]
g.	More than 100,000				[]
How lo	ng did you have your prin	nar	y cre	dit ca	ard?	
a.	Less than 1 year		[]		
b.	1 – 2 years		[]		
C.	3 – 4 years		[]		
d.	More than 4 years		[]		
What is	s vour monthly spending v	with	ı vou	r pri i	marv	rcredit card?
			,	ſ		
b.				ſ	_	
				ſ	-	
				ſ	-	
				ſ	_	
				٠	•	
		our	prin	nary	credi	it card?
		[]			
b. RN	/B201 – 300	[]			
C. RN	ИВ 301 – 400	[]			
d. M	ore than RMB401	[]			
e. Do	on't Know	[]			
What a	re the major activities for	VOI	ur pr	imar	v cre	edit card uses? (check all that apply)
a.	-	•	•	ſ	1	
b.				ſ	1	
c.	Entertainment			[]	
d.	Hotel Accommodation			ſ	1	
e.	Dining out			ĺ	1	
f.	_			ſ	1	
	Transfer accounts			1	1	
g.				•	-	
g. h.		one	dary	card	holde	er
g. h.	Facilitate the use of sec and to control the ex		•			
h.	Facilitate the use of sec	pen	ise of	fseco	onda	
h. i.Ot	Facilitate the use of sec and to control the ex her (please specific)	pen	ise of	f seco	ondai -	ry card holder [] []
h. i.Ot	Facilitate the use of sec and to control the ex her (please specific)	pen	ise of	f seco	ondai -	ry card holder []
	c. d. e. f. g. How lo a. b. c. d. What is a. Le b. RN d. Me. Do what a a. b. c. d. e. c. d. e. e.	c. RMB 5,001 to 10,000 d. RMB 10,001 to 20,000 e. RMB 20,001 to 50,000 f. RMB 50,001 to 100,000 g. More than 100,000 How long did you have your print a. Less than 1 year b. 1 – 2 years c. 3 – 4 years d. More than 4 years What is your monthly spending was a. Below RMB1000 b. RMB1001 to 5000 c. RMB5001 to 10000 d. RMB10001 to 20000 e. Above RMB20001 What is average annual fee on yas. Less than RMB 200 b. RMB201 – 300 c. RMB 301 – 400 d. More than RMB401 e. Don't Know What are the major activities for a. Shopping b. Online payment c. Entertainment d. Hotel Accommodation e. Dining out	c. RMB 5,001 to 10,000 d. RMB 10,001 to 20,000 e. RMB 20,001 to 50,000 f. RMB 50,001 to 100,000 g. More than 100,000 How long did you have your primar a. Less than 1 year b. 1 – 2 years c. 3 – 4 years d. More than 4 years What is your monthly spending with a. Below RMB1000 b. RMB1001 to 5000 c. RMB5001 to 10000 d. RMB10001 to 20000 e. Above RMB20001 What is average annual fee on your a. Less than RMB 200 b. RMB201 – 300 C. RMB 301 – 400 d. More than RMB401 e. Don't Know [What are the major activities for your a. Shopping b. Online payment c. Entertainment d. Hotel Accommodation e. Dining out	c. RMB 5,001 to 10,000 d. RMB 10,001 to 20,000 e. RMB 20,001 to 50,000 f. RMB 50,001 to 100,000 g. More than 100,000 How long did you have your primary cre a. Less than 1 year [b. 1 – 2 years [c. 3 – 4 years [d. More than 4 years [what is your monthly spending with you a. Below RMB1000 b. RMB1001 to 5000 c. RMB5001 to 10000 d. RMB10001 to 20000 e. Above RMB20001 What is average annual fee on your prima. Less than RMB 200 [b. RMB201 – 300 [c. RMB 301 – 400 [d. More than RMB401 [e. Don't Know [] What are the major activities for your prima. Shopping b. Online payment c. Entertainment d. Hotel Accommodation e. Dining out	c. RMB 5,001 to 10,000 d. RMB 10,001 to 20,000 e. RMB 20,001 to 50,000 f. RMB 50,001 to 100,000 g. More than 100,000 How long did you have your primary credit ca a. Less than 1 year [] b. 1 – 2 years [] c. 3 – 4 years [] d. More than 4 years [] What is your monthly spending with your primary a. Below RMB1000 [] c. RMB1001 to 5000 [] c. RMB5001 to 10000 [] d. RMB10001 to 20000 [] e. Above RMB20001 [] What is average annual fee on your primary a. Less than RMB 200 [] b. RMB201 – 300 [] C. RMB 301 – 400 [] d. More than RMB401 [] e. Don't Know [] What are the major activities for your primar a. Shopping [] b. Online payment [] c. Entertainment [] d. Hotel Accommodation []	c. RMB 5,001 to 10,000

	c.	6 to 10	%			[]								
	d.	More t	han 10%					[]						
11.	Is the	card you u	se most o	ften (prim	arv)	a Go	ld Ca	ard?						
	a.	Yes	[]	b.	No	u. y ,	1 1	1							
	۵.		. ,	٠.				,							
12.	How o	ften do yo	u use you	r crec	lit caı	rds fo	or pu	rcha	ses?						
	a.	rarely (few times	a yea	ar)					[]				
	b.	occasio	nally (at l	east c	nce a	a mo	nth)			[]				
	c.	freque	ntly (at lea	st on	ce a v	week	<)			[]				
	d. very frequently (almost daily) []														
13.	Wł	nen you us	e your cre	dit ca	ard, d	о уо	u aln	nost	alwa	ys pa	ay the	e total	balanc	e ea	ach
	mo	onth to avo	oid a finan	ce ch	arge?)									
	a.	Pay full a	mount				[]							
	b.	Sometim	es pay full	amo	unt		[]							
	C.	Hardly ev	er pay ful	l amo	unt		[]							
14.	Wł	nat is the n	nain reasc	n you	u do i	not p	oay th	ne fu	II am	oun	t of tl	ne tota	al balan	ce c	on
	you	ur monthly	credit ca	rd bill	?										
	a.	Lack of	fund to p	ay ba	ck at	once	е						[]	
	b.	Bank al	lows insta	llmer	nt pay	ymei	nt						[]	
	с	There i	s no fees o	or inte	erest	occı	urred	with	n inst	tallm	ent p	ayme	nt []	
	d.	Other (please spe	ecify)	_								[]	
15.	How lo	ong will it t	ake you to	o pay	off y	our d	credit	card	ds ba	lanc	es?	I carr	у		
	a.	no bala	ince						[]					
	b.	1-3 mo	nths						[]					
	c.	4-6 mo	nths						[]					
	d.	7-12 m	onths						[]					
	d.	more t	han 1 year	•					[]					
16.	Do you	u use your	credit car	ds to	make	e onl	ine p	aym	ents	?					
	a.	Yes	[]	b. N	No		[]							
17.	How o	often do yo	u use you	r prin	nary	cred	it car	d fo	r Inte	ernet	(on-	line) p	urchase	es?	
	a.	rarely (few times	a yea	ar)						[]			
	b.	occasio	nally (at l	east c	once a	a mo	nth)				[]			
	c.	freque	ntly (at lea	ist on	ce a	weel	k)				[]			
	d.	very fre	equently (almos	st dai	ly)					[]			

Section 3. Non Credit Card Holders

1. Why don't you have a credit card?	(Check all ti	nat a	pply)	
a. Inconvenient			[]
b. Application fee charges are	too high		[]
c. Annual fee charges are too h	nigh		[]
d. The fees and service charges	s are too hig	h	[]
e. I worry the risk of accumulat	ting debts		[]
f. I am unable to quality	[]		
g. I have security concerns	[]		
h. I have privacy concerns	[]		
i. I have difficulty record keeping	[]		
j. I don't need a credit card	[]		
k. Other (please specify)			[]
 If a bank or credit card company o 	offered you a	crac	lit ca	rd at no cost to you would
you be more likely to use the credi			iit ca	ru, at no cost to you, would
	Credit card		[]
3. Do you intend to have a credit card?				
a. Yes [] b. No []	c. Not sure	[]	
(check all that apply) a. No annual fee b. Preapproved c. Rebate/cash back/free offer d. Business/corporate card e. Gold Card f. High credit limit g. Wide acceptance (include or h. Lower interest rate i.Reward points program j.Billing cycle/grace period k. Lower monthly payment	[[nline shoppi [[[]]	[]
I. Other (please specify)		[]	
Section 4. Demographic and Soci	o-economic	Char	acte	ristics of Respondents
· .	l Responden			
·	-	•		
1. What is your gender?				
	nale []		

2.	Whic	ch age group do you belong to?					
	a.	18 – 25 years old	[]			
	b.	26 – 35 years old	[]			
	c.	36 – 45 years old	[]			
	d.	46 – 55 years old	[]			
	e.	56 – 65 years old	[]			
	f.	Over 66 years old	[]			
3.	Wha	t is your marital status?					
	a.	Single/Never Married	[]			
	b.	Married	[]			
	С.	De factor relationship	ſ	1			
	d.	Divorced/Separated	[]			
4.	Wha	t is your highest educational or profession	al nu	alific	ation	17	
••	a.	No education	ی. و ا	1			
	b.	Primary school	[1			
	С.	Middle school	ſ	1			
	d.	High school	ſ	1			
	e.	Three-year college	ſ	1			
	f.	Bachelor degree	ſ	1			
	g.	Postgraduate degree (Postgraduate Diplo	ι nma/	Mas	ters/	PhD)	[]
	h.	Other (please specify)	Ja,	11100	[]	. ,
					٠	•	
5.	Whi	ch of the following best describes your em	ployr	nent	situa	ation?	
	(If a	or b, please go to Q6, if c, d, e, f, or g, pleas	se go	to C	(<i>7)</i>		
a.	Full t	ime		[]		
b.	Part						
c.		time		[]		
	Uner	time nployed		[[]		
d.]]]]]		
		mployed employed]]]]]]]		
	Self- Retir	mployed employed]]]]]]]]]		
e.	Self-G Retir Migr	mployed employed ed]]]]]]]] []	
e. f.	Self-G Retir Migr Othe	mployed employed ed ant worker	er, wh	[[[[—]]]] [ing best
e. f. g.	Self-G Retir Migr Othe	mployed employed ed ant worker er (please specify)	er, wh	[[[[—]]] [of the		/ing best
e. f. g. 6.	Self-c Retir Migr Othe	mployed employed ed ant worker er (please specify) u work as permanent staff or casual worke	er, wh	[[[]]] [of the		ing best
e. f. g. 6.	Self-G Retir Migr Other If you desc Gove	mployed employed ed ant worker er (please specify) u work as permanent staff or casual worker	er, wh				ing best
e. f. g. 6.	Self-G Retir Migr Othe If you desc Gove State	employed employed ed ant worker er (please specify) u work as permanent staff or casual worker eribes your employer?	er, wh				ing best
e. f. g. 6. c.	Self-G Retir Migr Other If you desc Gove State Priva	employed employed ed ant worker er (please specify) u work as permanent staff or casual worker ribes your employer? ernment office e-owned enterprise	er, wh				ing best
e. f. g. 6. c. d.	Self-GRetin Migro Other If you describe States Private Foreign Self-GRETING States Foreign Self-GRETING Self-	employed employed ed ant worker or (please specify) u work as permanent staff or casual worker ribes your employer? ernment office e-owned enterprise or enterprise	er, wh				ving best

	g. No	n-governmental organisation	[]				
	h. O	ther (please specify)		[]			
7.	How long have you been working (total working period: past and/or current job							
	a. Les	s than 1 year	[]				
	b. 1 to	5 years	[]				
	c. 5 to	10 years	[]				
	d. 10 y	rears and above	[]				
8.	Which	of the following best describes the structure of	of your	hous	sehold?			
	a.	Single adult living alone			[]			
	b.	Couple, with child (or children)			[]			
	c.	Couple, without child			[]			
	d.	Single parent, with child (or children)			[]			
	e.	Immediate and extended family members			[]			
	f.	Other(s) please specify			_[]			
9.		mber of people living in your household is (pla persons	ease sta	ate):				
10.	The nu	mber of income earners in your household is (persons	(please	state	e):			
11.	What i	s your household monthly income? (Chinese R	MB in	the l	ast month)			
	a.	1,000 RMB or less	[]				
	b.	1,001 to 2,000 RMB	[]				
	C.	2,001 to 3,000 RMB	[]				
	d.	3,001 to 4,000 RMB	[]				
	e.	4,001 to 5,000 RMB	[]				
	f.	5,001 to 7,000 RMB	[]				
	g.	7,001 to 10,000 RMB	[]				
	h.	Above 10,000 RMB	[]				

Your participation in this survey is greatly appreciated. Thank you for your time and if you have further comments about credit card, please feel free to comment in the space provided below. Once again, we assure you that your identity will remain **STRICTLY CONFIDENTIAL.**